

Delft University of Technology

A review on application of machine learning-based methods for power system inertia monitoring

Heidari, Mahdi; Ding, Lei; Kheshti, Mostafa; Bao, Weiyu; Zhao, Xiaowei; Popov, Marjan; Terzija, Vladimir

DOI 10.1016/j.ijepes.2024.110279

Publication date 2024 **Document Version** Final published version

Published in International Journal of Electrical Power & Energy Systems

Citation (APA)

Heidari, M., Ding, L., Kheshti, M., Bao, W., Zhao, X., Popov, M., & Terzija, V. (2024). A review on application of machine learning-based methods for power system inertia monitoring. *International Journal of Electrical Power & Energy Systems*, *162*, Article 110279. https://doi.org/10.1016/j.ijepes.2024.110279

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

Contents lists available at ScienceDirect



International Journal of Electrical Power and Energy Systems

journal homepage: www.elsevier.com/locate/ijepes



Review

A review on application of machine learning-based methods for power system inertia monitoring

Mahdi Heidari^a, Lei Ding^{a,*}, Mostafa Kheshti^b, Weiyu Bao^a, Xiaowei Zhao^b, Marjan Popov^c, Vladimir Terzija^d

^a Key Laboratory of Power System Intelligent Dispatch and Control (Shandong University), Ministry of Education, Jinan 250061, China

^b Intelligent Control and Smart Energy (ICSE) Research Group, School of Engineering, University of Warwick, Coventry CV4 7AL, UK

^c Delft University of Technology, Faculty of EEMCS, Mekelweg 4, 2628CD Delft, the Netherlands

^d Newcastle University, School of Engineering, Newcastle Upon Tyne NE1 7RU, UK

ARTICLE INFO

Keywords: Converter interfaced generation Frequency stability Inertia monitoring Machine learning Synchronous generators System inertia

ABSTRACT

The modernization of electrical power systems is reflected through the integration of renewable energy resources, with the ultimate aim of creating a carbon–neutral world. However, this goal has brought new and complex challenges for the power system, with one of the most crucial issues which is the reduction of system inertia. The decrease in system inertia has led to severe difficulties in maintaining frequency stability. As a result, power system operators must continuously monitor the system inertia and when necessary to activate appropriate preventive measures, ensuring a reliable and secure operation of the power system. Fortunately, wide-area monitoring systems can provide the necessary measurements to monitor and analyze system behavior, assisting system operators in undertaking optimal actions. This paper provides a review of recent publications that apply machine learning (ML)-based methods for monitoring power system inertia. It also provides an overview of academic and industrial projects related to ML-based methods for inertia monitoring. Furthermore, the paper explores applications based on ML-based methods and inertia. Lastly, the paper briefly discusses future directions for the development of this research field.

© 2017 Elsevier Inc. All rights reserved.

1. Introduction

To reduce dependency on fossil fuels and curb greenhouse emissions, power systems started undergoing a systematic integration of renewable energy sources (RESs) [1]. In response to this ultimate need, traditional power plants, primarily relying on synchronous generators (SGs), are being massively replaced by converter-interfaced generation (CIG), in particular wind turbine generators (WTGs) and photovoltaic (PV) plants. Furthermore, the introduction of power electronics-based devices in both transmission and consumption aspects, such as highvoltage direct current (HVDC) transmission lines, battery energy storage systems (BESSs), and converter-connected motors, is transitioning the entire power system from its conventional mode to a new one, dominated by converters, as depicted in Fig. 1.

One of the primary challenges stemming from the system transformation is the reduction in system inertia, a concern increasingly prevalent in power systems worldwide [2]. In traditional power systems, inertia is provided predominantly by the rotational inertia of SGs and directly-coupled motors, which act to account for deviations in grid frequency and maintain stable frequency profiles. However, as RESs replace SGs, e.g. PV plants, BESSs, and WTG plants (Type 3 - Doubly Fed Induction Generator (DFIG) and Type 4 - Full-Scale Converter (FSC) wind turbine generators (WTGs)), the overall system inertia declines. The presence of converter-connected motors exacerbates this reduction. Consequently, the power system experiences a higher rate of change of frequency (RoCoF) and extreme frequency variations following disturbances, as shown in Fig. 2. These conditions can trigger load shedding schemes, activate other protection systems, and lead to subsequent cascading events, potentially culminating in blackouts [3]. As a result, concerns regarding frequency stability have come to the forefront, playing a critical role in ensuring reliable operation of power systems. In the past decade, numerous blackout incidents were reported and globally investigated; major events that have drawn the attention were in the United Kingdom [4], China [5], Australia [6], India [7], and

https://doi.org/10.1016/j.ijepes.2024.110279

Received 29 January 2024; Received in revised form 31 August 2024; Accepted 30 September 2024 Available online 22 October 2024

^{*} Corresponding author. *E-mail address:* dinglei@sdu.edu.cn (L. Ding).

^{0142-0615/© 2024} The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

the system split event in Europe [8]. These incidents were primarily attributed to insufficient system inertia.

In addition to the decline in system inertia, there is a fundamental shift in the nature of inertia itself. Formerly a steady-state parameter, inertia is now characterized by time-varying and spatial features. The number of online synchronously operating SGs, responsible for contributing to system inertia, fluctuates due to the intermittent nature of the RESs. In response, inertia changes dynamically with the variation of weather conditions at RES plants, including factors such as wind speed, irradiation, and temperature [9]. Furthermore, the concept of virtual inertia, supported by RESs, is subjected to variability depending on the operational limits of these resources. A prime example is observed with WTGs when employed to support system inertia. In such cases, the supported inertia must be finely adjusted to ensure the stability of WTGs, particularly in terms of their variable rotor speed, during their exploitation in the power system [10]. Moreover, the power system undergoes partitioning and separation into multiple regions facilitated by power converter-based devices like HVDC transmission lines. Consequently, SGs provide inertia within specific regions, leading to distinct frequency dynamics across various parts of the power system [11].

Due to these complex conditions, system operators face difficulties in making decisions related to dispatching and implementing rapid frequency response plans defined as Fast Frequency Control service. The challenges that have emerged underscore the importance of precise and near-real-time monitoring of system inertia in safeguarding the reliability and security of the power system. Monitoring system inertia can offer valuable reference inputs for protection and proactive control systems as the impacts of system inertia on the frequency stability is shown in Fig. 2. This approach not only boosts system stability but also improves operational efficiency. For instance, it enables more informed planning of ancillary reserves by leveraging insights into current system conditions and stability margins [12]. Research in the field of inertia monitoring is actively progressing, with a rich literature presenting various methods to address this need.

Machine learning (ML) emerges as a highly effective solution to address a number of complex problems. ML's capability to analyze vast volumes of multi-variety data allows it to uncover specific trends and patterns. These methods continuously improve their accuracy and efficiency through experience, without requiring human intervention [13]. Recently, ML-based methods have found extensive application in power system assessment, particularly through the use of wide-area monitoring systems [14]. Applications include reliability assessment [15], analysis of frequency dynamics [16,17], and resilience assessment [18].

In contrast to existing papers that review some methods for monitoring system inertia [12,19-22] and the significance of ML-based applications in power system analysis and control, this review aims to digest and summarize key insights from existing references found in the open literature as well as industry projects for monitoring system inertia based on ML-based methods. While there have been limited instances of joint applications of system inertia monitoring and ML-based algorithms in the existing literature, this paper intends to showcase these applications with the aim of inspiring further research and exploration in this field. Furthermore, it attempts to chart a path for future developments in this emerging field of engineering by identifying knowledge gaps, forthcoming requirements, and possible solutions. The primary contributions of this work are as follows:

- Review of system inertia definitions and approaches for its monitoring in both traditional and modern power systems.
- Investigation of ML-based methods used for inertia monitoring, providing an overview for researchers interested in developing new methods in this field or other related applications, e.g., frequency control.
- Comprehensive and comparative review of system inertia monitoring using ML-based methods, including an analysis of various scenarios for data collection and inertia monitoring.
- Review of academic and industrial projects related to system inertia monitoring using ML-based methods.
- Review of the limited literature on applications involving ML-based methods and associated inertia monitoring, aiming to inspire experts to further explore this cutting-edge field.
- Provision of future directions by taking into account gaps in existing published open literature, that will serve as a roadmap for further research in this field.

The paper is structured as follows: Section 2 focuses on defining system inertia, followed by a review of ML-based methods applied to system inertia monitoring in Section 3. Section 4 explores inertia monitoring methods employing ML-based techniques found in the literature, while Section 5 presents comprehensive academic and industrial projects related to system inertia monitoring using ML-based methods. Section 6 highlights applications that utilize ML-based methods and determined inertia for different types of applications like power system analysis, control, and protection. Finally, Section 7 presents the conclusion and outlines future directions.

2. Definition of system inertia

Conventionally, system inertia is defined as resistance to the speed changes of the SGs' rotors [23]. The moment of inertia (MoI) is the



Fig. 1. Transition of the power system from a traditional one (synchronously operating generators) to a new one (converter-interfaced generation - CIG).

quantitative measure of rotational inertia for a rigid body rotating around an axis. Let's consider a cylinder with the mass of m rotating around the *x*-axis as shown in Fig. 3. The MoI of this cylinder according to the *x*-axis is defined as follows:

$$J_{\text{cylinder}} = \frac{1}{2}mr^2 \tag{1}$$

where *m* and *r* are the mass in kg and radius in m of the cylinder. $J_{cylinder}$ is the MoI of the cylinder expressed in kgm^2 . To determine the MoI of SG's rotor with the multi-mass objects that are rotating, firstly, the MoI of each rotating component of the rotor including core packs, copper wire, insulations, wedges, bearings and fan is calculated concerning its center of gravity [24]. Then, these MoIs are modified according to the distance from their centers of gravity to the main rotor centerline by using the parallel axis theorem [25]. Finally, the MoI of SG's rotor is the sum of the MoI of all rotating components as,

$$J = \sum_{i} (J_{c_i} + m_{c_i} * r_{c_i})$$
(2)

J is the MoI of SG's rotor and J_{c_i} is the MoI of *i*-th rotating component due to its center of gravity. m_{c_i} is the mass of *i*-th rotating component and r_{c_i} is the distance between the centerline of the rotor and the center point of *i*-th rotating component. The kinetic energy stored in the rotating masses of the SG's rotor is defined as,

$$E_k = \frac{1}{2} J \omega_m^2 \tag{3}$$

 E_k is the kinetic energy of the SG's rotor in MVAs; ω_m is the angular velocity of SG in *rad/s*. When a torque mismatch occurs for torques acting on the rotor of a SG, the SG naturally releases or absorbs its kinetic energy stored in the rotating masses. This process is called inertial response and could be described mathematically by the swing equation of the SG in terms of mechanical and electrical powers [26],

$$\Delta P_{m,i} - \Delta P_{e,i} = 2H_i \frac{d}{dt} \left(\Delta \omega_{m,i} \right) \tag{4}$$

 $\Delta \omega_{m,i}$ is the deviation of the angular velocity of *i*-th SG from nominal velocity in per unit (p.u.). $\Delta P_{m,i}$ is the change of mechanical power input of *i*-th SG in p.u. corresponding to the power supplied by it. $\Delta P_{e,i}$ is the change of electrical power output of *i*-th SG in p.u. related to the power demand from the load. If the damping factor of frequency-dependent loads is included, the aforementioned swing equation is modified according to [26]:

$$\Delta P_{m,i} - (\Delta P_{L,i} + D_i \Delta \omega_{m,i}) = 2H_i \frac{d}{dt} \left(\Delta \omega_{m,i} \right)$$
(5)

 D_i is the damping factor and $\Delta P_{L,i}$ is the power change of frequencyindependent loads in p.u. H_i is the inertia constant of *i*-th SG in sec-



Fig. 3. Cylinder rotating around the x-axis.

onds and is defined as [26],

$$H_{i} = \frac{E_{k_{i}}}{S_{r_{i}}} = \frac{\frac{1}{2}J_{i}\omega_{m_{0},i}^{2}}{S_{r_{i}}}$$
(6)

 E_{k_i} , S_{r_i} , $\omega_{m_0,i}$, and J_i are the kinetic energy in MVAs, the rated apparent power (base power) in MVA, the rated rotor speed, and the rotor's MoI of *i*-th SG, respectively. The amount of kinetic energy released or absorbed by each SG depends on the MoI and the speed $\omega_{m_0,i}$. Consequently, the rotor speed and, by extension, the system frequency, will vary in proportion to the MoI. In power system studies, the inertia constant, H, is commonly used to express the system's ability to resist changes in rotational speed. The inertia constant has to be understood as a time constant. The physical significance of H lies in the fact that when the input mechanical power of an SG is zero, and SG injects nominal electrical power solely by releasing its stored kinetic energy, the rotor speed of the SG will reach zero after 2H seconds [26]. In other words, within this time frame, the SG expends all of its stored kinetic energy to compensate for the active power imbalance, equal to its nominal power, without any mechanical power input. When equivalenting the entire electrical power system with a single generator, it is possible to define the system inertia constant, H_{sys} , and frequency of the equivalent inertia center, f_{Col} , [27,28]. This can be realized by summing up the H_i values of all individual SGs and considering their measured frequencies, power system inertia and the frequency of the center of inertia can be expressed as:

$$H_{\rm sys} = \frac{\sum_{i=1}^{N_{SC}} S_{r_i} H_i}{\sum_{i=1}^{N_{SC}} S_{r_i}}$$
(7)

$$f_{CoI} = \frac{\sum_{i=1}^{N_{SG}} H_i f_i}{\sum_{i=1}^{N_{SG}} H_i}$$
(8)

where H_{sys} is the system inertia constant in seconds and f_{Col} is the frequency of the center of inertia in Hz. f_i is the frequency of the *i*-th SG and N_{SG} is the number of online SGs. Similarly, the aggregated swing



Fig. 2. Effects of system inertia decline on the frequency deviations following a disturbance event.

equation can be approximated by ignoring the damping factor in the initial moments of disturbance as follows:

$$\frac{2H_{sys}}{f_n} \times \frac{df_{CoI}}{dt} = \frac{P_{generation} - P_{load}}{S_B} = \frac{\Delta P}{S_B}$$
(9)

where $P_{generation}$, P_{load} , and ΔP are the aggregated generated power, power demand, and active power mismatch in MVA, respectively. f_n is the nominal frequency in Hz and S_B is the base power in MVA. The system RoCoF is obtained from (9):

$$RoCoF = \frac{df_{Col}}{dt} = \frac{\Delta P \times f_n}{2 \times H_{sys} \times S_B}$$
(10)

It is evident from (10) that the frequency dynamics of the power system dominated by SGs highly depend on the system inertia. In modern power systems penetrated by high shares of RESs, system inertia is provided from two kinds of sources: rotational inertia resulting from synchronous machines and virtual inertia supported by RESs [29],

$$H_{sys} = \frac{\sum_{i=1}^{N_{SG}} S_{r_i} H_i + \sum_{j=1}^{N_{VI}} S_{VI_j} H_{VI_j}}{S_B}$$
(11)

 S_{VI_j} and H_{VI_j} are the nominal apparent power and virtual inertia of *j*-th RES contributing to the virtual inertia, respectively. N_{VI} is the number of RESs supporting the virtual inertia. According to this equation, the system inertia is not determined by the number of online SGs but is obtained by considering the frequency variations and active power mismatch between generation and consumption based on (9).

3. Ml-based methods applied to inertia monitoring

To introduce the ML-based methods employed in system inertia monitoring, an overview of common terminologies is made. This overview outlines the core of ML-based problems, and explain the structure of ML-based methods. The fundamental objective behind developing an ML-based method is to make predictions or estimations based on the input features or measurements. In our case, this estimation capacity pertains to determining the system inertia.

To fulfill this objective, ML-based methods are trained using a dataset. Depending on the nature of the target and training data, MLbased methodologies can be broadly classified into three main categories: supervised learning, unsupervised learning, and reinforcement learning (RL) [30]. In supervised learning, model is trained iteratively, relying on labeled datasets. Model's primary goal is to minimize the difference between its estimations and the provided correct answers. Common ML models for supervised learning include decision trees [31], neural networks [32,33], linear regression [34], nearest neighbor [35], and support vector machines [36,37]. In our context, most of the system inertia monitoring methods based on ML-based methods fall under supervised learning, with the system inertia value serving as the label. In contrast, unsupervised learning operates without labeled data. Methods in unsupervised learning autonomously explore and discern inherent structure within unlabeled datasets, uncovering patterns, groupings, or relationships within the data. The K-means algorithm [38], principal component analysis [39], and hidden Markov models [40] are the most famous models for unsupervised learning.

RL represents a distinct learning approach, setting itself apart from the two previously mentioned methods. RL focuses on how an agent makes sequential decisions within an uncertain environment to maximize cumulative rewards. Its applications in power system operation and control have recently garnered significant attention [41,42]. Given the advanced decision-making capabilities inherent in RL, researchers have integrated it with the perceptual prowess of Deep Learning (DL), culminating in the Deep Reinforcement Learning (DRL) framework [43]. The various advantages of DRL techniques, including a high-complexity processing system, data-driven and model-free approaches, and adaptability, have led to endeavors in recent years to apply DRL in wind farm control tasks [44,45], including wind farm power regulation and tracking [46,47]. Moreover, DRL finds applications in diverse areas such as power market trading [48], frequency response [49], demand response [50], and autonomous voltage control [51].

Fig. 4 illustrates the typical structure involved in constructing an MLbased method, showcasing the different stages of dataset building, data preprocessing, model training, and evaluation. The initial step in building an ML-based method is the creation of a robust dataset. In cases where historical data may be scarce, simulations are conducted to generate a training dataset. Striking the right balance in the dataset's representation is pivotal for achieving a reliable and practical ML-based method.

After constructing a dataset, the next important step is the data preprocessing before it is utilized by a learning algorithm. During this stage, feature engineering plays a pivotal role, encompassing both feature extraction and feature selection techniques. Feature extraction involves transforming the data to represent it in a more meaningful and compact manner. This transformation serves to enhance the model's ability to capture essential patterns within the data while reducing its dimensionality [52]. Feature selection algorithms such as wrapper and genetic algorithms are employed to identify and retain the most relevant and informative features from the dataset [53]. Focusing on data key aspects, feature selection simplifies the model, mitigates the risk of overfitting, and ultimately leads to more accurate and efficient learning.

Numerous learning algorithms are available for constructing MLbased methods, and the choice of the most suitable one depends on various factors like the type of dataset and the specific requirements of the task. Typically, the training takes place offline, but methods may also be updated online as new data becomes available. Given the diverse landscape of the learning algorithms, a prudent approach involves experimenting with different learning algorithms and assessing their performance against the dataset.

The final stage of ML-based method construction is model validation, a crucial step in the development of any ML-based system. By testing the model on unseen data, it verifies the generalization capability of the established model. With these considerations, the subsequent sections will delve into the specific models applied to ML-based system inertia monitoring methods, paying attention to their methodologies and applications.

3.1. Artificial neural network

In the various ML-based techniques, Artificial Neural Network (ANN), inspired by the intricate workings of the human brain, has garnered significant attention across a wide array of research domains [54]. An ANN is a computational model composed of interconnected nodes (neurons) arranged in layers: an input layer, one or more hidden layers, and an output layer. This layered architecture is depicted in Fig. 5. Each connection between nodes has an associated weight that is adjusted during training. The network learns by optimizing these weights to minimize the difference between predicted and actual outputs. The main advantages of ANN are including versatility, learning capability, non-linearity, feature extraction, and scalability. While, the drawbacks of this ML model consist of data requirements, computational resources, interpretability, overfitting, and hyper-parameter tuning. ANN is particularly suitable for static data scenarios, where the same conditions are repeatable, such as in pattern recognition [55], classification [56,57], forecasting [58,59], and similar tasks.

3.2. Recurrent Neural Network

Recurrent Neural Network (RNN) is a specialized type of neural networks designed to handle time series data or sequences, which can originate from various sources such as text, speech, or video. Fig. 6 provides an overview of the inner workings of an RNN unit, which is



Fig. 4. General structure of constructing an ML-based method.



Fig. 5. Typical structure of MLP.



Fig. 6. Structure of an RNN unit.

central to its sequential processing. One of the key attributes that sets RNNs apart is their ability to incorporate a form of "memory", allowing them to retain and utilize information from prior inputs, which influences the processing of future time steps to generate subsequent outputs within a sequence [60]. The main advantages of RNN are including sequential data handling, temporal dependencies, variable input/output length, and parameter sharing. However, the drawbacks of this ML model consist of vanishing gradients, training difficulty, limited memory, computational cost, and complexity. To address the vanishing gradient problem, the Long Short-Term Memory (LSTM) network has emerged as a remarkable RNN architecture. LSTMs are designed specifically to overcome this challenge, enabling the network to capture and utilize information over long sequences effectively [61]. The impact of LSTM-based networks has been profound across various fields, including language modeling [62], speech-to-text transcription [63], machine translation [64], and other applications.

3.3. Convolutional neural network

Convolutional Neural Network (CNN) is a widely used class of neural networks, especially for spatial patterns analysis. CNNs are specifically designed to learn spatial features within the data, such as edges, corners, textures, or more complex shapes, that are essential for classifying or quantifying targets [65]. A typical CNN architecture, as shown in Fig. 7, consists of several key layers, including convolutional layer for extracting feature map, pooling for down-sampling feature maps, flatten layer for reshaping the data into one-dimensional array, and fully connected layer for classification. The main advantages of CNN are including automatic feature extraction, spatial hierarchy, parameter sharing, and scalability. While, the drawbacks of this ML model consist of computational resources, interpretability, overfitting, design complexity, lack of interpretability, and need for large datasets. CNNs find their applications in various real-world problems, including but not limited to image recognition [66,67], natural language processing [68], video analysis [69], anomaly detection [70], drug discovery and health risk assessment [71], and recommender systems [72]. There are several modified architectures of CNNs, some of which have been adapted for monitoring system inertia. These models support the core principles of CNNs while incorporating specific modifications to address the unique challenges of system inertia monitoring.

3.3.1. Graph Convolutional network

Graph Convolutional Network (GCN) is a combined architecture of CNNs and graph neural networks (GNNs) models specially designed for processing and making inferences on graph-structured data, where nodes represent entities and edges represent relationships between them [73]. One of the key strengths of GCNs is their ability to aggregate node information from neighbourhoods in a convolutional manner. This enables the model to learn and represent complex relationships between nodes within the graph. GCNs have found applications in various domains, including making recommendations in the recommender systems [74,75], and forecasting traffic conditions [76]. In the context of power systems, which consist of interconnected generators and loads, GCNs are applied to understand the graph structure of the power systems, comprising nodes (buses) and edges (branches) [77,78]. Another advanced model of GNNs used in the system inertia monitoring is Graph Attention Networks (GATs). GATs allocate different levels of attention to neighbouring nodes during information diffusion, learning the



Fig. 7. Typical CNN architecture.

importance of each node in a graph [79]. By utilizing attention mechanisms, GATs effectively capture intricate relationships and spatial dependencies, improving their performance in tasks like node classification and link prediction.

3.3.2. Residual neural network

Residual neural Network (ResNet) represents a significant advancement in CNN-based architectures, addressing the vanishing gradient problem that tends to occur in deep networks with many layers [80]. ResNet introduces a "shortcut connections" or "skip connections" to facilitate the training of very deep networks, allowing the gradients to flow more effectively through the network without vanishing or exploding. The ability to train very deep networks with ResNet has led to improved accuracy on various computer vision tasks, such as image classification [81], object detection [82], and segmentation [83].

3.3.3. Long-Term Recurrent Convolutional network

Long-term Recurrent Convolutional Network (LRCN) is a hybrid neural network architecture that combines the strengths of CNN and LSTM networks [84] to integrate spatial and temporal learning. The strength of the LRCN lies in its ability to integrate the spatial features extracted by the CNN with the temporal context captured by the LSTM, making it particularly well-suited for applications like activity recognition [85], image captioning [86], and video description.

3.4. Physics-Informed neural network

Physics-Informed Neural Network (PINN) represents a novel approach to solving problems with limited data, particularly when dealing with noisy experimental measurements. These networks are designed to seamlessly integrate information from both data measurements and the governing physical laws of the problem [87]. PINNs do this by embedding Partial Differential Equations (PDEs) into the loss function of a neural network using automatic differentiation. Fig. 8 shows the general structure of a PINN. The capability of PINN in addressing the complex problems that involve physical laws and limited data has led to gain significant attention and exploration in various scientific and engineering disciplines due to its potential advantages, including hydrology, chemistry, materials science, earth systems, hydromechanics, and more [88–91] as well as power system studies [92,93].

3.5. Federated learning

Federated Learning (FL) represents a significant shift in the field of ML, transitioning from a centralized and resource-intensive approach to a distributed paradigm that leverages the power of numerous distributed computing resources [94]. This approach not only offers advantages in terms of data privacy but also reduces network overhead by transmitting



Fig. 8. General structure of a PINN.

only model updates rather than raw data [95]. FL, as illustrated in Fig. 9, involves two main entities: clients and a central server. Clients typically represent individual devices or data sources, while the central server coordinates the learning process. In recent years, FL has demonstrated successful applications across various industries, particularly those related to the Internet of Things and edge computing, where privacy-preserving ML-based methods are crucial [96]. Also, it is gaining interest in high-dimensional organizations, energy systems, and smart grids with distributed data sources, where data privacy and security are paramount [97].

3.6. Multivariate random Forest regression

Multivariate Random Forest Regression (MRFR) is a powerful supervised ML-based algorithm that leverages the principle of random forests for regression tasks. Fig. 10 shows the typical structure of the MRFR algorithm. MRFR is an ensemble learning technique that builds multiple base models (regression trees) and aggregates their predictions to make accurate and robust predictions [98]. It has been successfully applied in various domains, including disease prediction [99], mobile networks [100], localization systems [101], and parameter assessment and estimation [102]. Table 1 summarize the main features, variants, pros and cons of ML models used for ML-based system inertia monitoring methods.

4. Ml-based methods for inertia monitoring

Implementation of ML-based methods for inertia monitoring in power systems involves several stages. These stages, illustrated in the typical process depicted in Fig. 11, include data acquisition, feature engineering, model training and validation, and the online deployment of the trained model. In the context of data acquisition and the scenarios under which data is collected, inertia monitoring methods based on MLbased algorithms can be broadly classified into three main methods: contingency operation-, normal operation-, and perturbation-based approaches. These methods are discussed here below.

4.1. Contingency Operation-Based methods

In these methods, system's reactions during the events are analyzed by an ML-based algorithm to monitor the system inertia. To train and validate the ML-based method, the size and type of disturbances, as well as the system inertia, are varied, and a variety of measurements are collected. The extracted inertia information can be used for designing load shedding plans [103] and fast frequency control schemes for RESs [104,105].

The paper in [106] presents an ANN-based method for inertia monitoring using measured system frequency signal as the input. This approach involves discretizing the frequency signal and feeding each



Fig. 9. Diagram of FL structure.



Fig. 10. Typical architecture of the MRFR algorithm.

Table 1 ML mod

	L	models	used	for	ML	-based	system	inertia	monitoring	methods
--	---	--------	------	-----	----	--------	--------	---------	------------	---------

Model	Main features	Variants	Pros/Cons
ANN	Computational model composed of interconnected neurons arranged in layers.		 + Versatility; learning capability; feature extraction; non-linearity; scalability. - Data requirements; interpretability; computational resources; overfitting; hyper-parameter
RNN	Well-suited for tasks where data is sequential or time- dependent.	LSTM	 tuning. + Sequential data handling; temporal dependencies; variable input/output length; parameter sharing. - Vanishing gradients; training difficulty; limited memory; computational cost; complexity.
CNN	Effective for processing the grid-like data, such as images.	GCN ResNet LRCN	 Automatic feature Automatic feature extraction; spatial hierarchy; parameter sharing; scalability. Computational resources; design complexity; interpretability; overfitting; lack of interpretability; need for large datasets
PINN	Incorporation of governing equations of a physical system in ML model.		 b) the integration of b) physics; data efficiency; flexibility; generalization; reduced computational cost. Training complexity; limited to well-defined physics; convergence issues; choice of network architecture; accuracy of physical models
FL	To allow multiple decentralized devices to collaboratively train a model; Keeping the data localized on each device.		 Puster indexts. Data privacy and security; reduced data transfer; scalability; adaptability; regulatory compliance Communication overhead in large models; computational resource requirements; process of heterogeneous data; model aggregation challenges;
MRFR	An ensemble learning technique; To build multiple decision trees and aggregates their predictions.		 Handling multiple targets; Handling multiple targets; robustness; feature importance; non-linearity; no need for scaling. Computational complexity; model interpretability; overhead of multiple targets; memory usage; hyper- parameter tuning.

frequency sample to the individual neurons in the input layer of the ANN. The dataset used for training and validation is generated by varying the system inertia and event size (both positive and negative). While this method was initially proposed and verified for a small power system, its practical applicability to larger-scale power grids was limited due to the system's size and complexity.

To address this limitation and make the inertia monitoring method more applicable to realistic conditions, researchers used the IEEE 39-bus test system [107–110]. The frequency signals from all buses in the IEEE 39-bus test system are measured [107] and these measurements are taken during various generator and load events, allowing for a diverse dataset. One crucial aspect highlighted in [107] is the careful selection of f_{COI} for achieving high accuracy in inertia monitoring. Both f_{COI} and its corresponding RoCoF are calculated from the measured frequency signals to train the ResNet model.

Frequency signal and active power imbalance are chosen as input features to train and test the ML-based methods [108,109]. Considering load and generator disturbances, a one-dimensional CNN is applied to establish the relation between f_{COI} and system inertia [108]. Since LSTM is a powerful tool analyze the data sequences, it is implemented in [109] by considering the load events. Because the active power mismatch is prevalently unknown, the selection of this parameter as the input feature of the model is a challenge which is highlighted in these two methods. Besides, the use of local frequency signals instead of f_{COI} may not capture the system-wide behavior accurately in [109].

The method proposed in [110] addresses the challenge in monitoring the regional distribution of system inertia in power systems with increasing shares of RESs. The primary goal of this method is to monitor the regional distribution of system inertia. Generators with similar frequency response characteristics are clustered using the K-means algorithm. The input features of the model, which is LSTM, are a wide range of data, including the measured signals of frequency of SGs' buses, active and reactive powers of SGs and loads, and voltage and phase angles of all buses. This comprehensive set of features allows the model to capture the spatial distribution of inertia by considering the data from various sources in the power system. However, clustering the generators accurately based on the frequency response characteristics can be challenging as the accuracy of the clustering process affects the model's performance. Additionally, the integration of RESs and their inherent uncertainty can introduce complexities that may need further consideration.

To optimize the computation time for monitoring the system inertia using ML-based methods, an autoencoder is employed in [111] to compress the large volume of data gathered from synchro-phasors through unsupervised learning. In this ML-based approach, locally measured signals of system frequency and RoCoF from selected buses are fed into the autoencoder. This process maps the extensive frequency data to a small, and finite number of abstract features-specifically, five attributes. These features are then used as inputs for an ANN to monitor the system inertia. The limitations of this proposed method include the use of a simple single-machine model of the power system for collecting the training dataset, the exclusion of noise in frequency measurements, and the lack of consideration for the participation of RESs in virtual inertia. Additionally, the performance of this method in large-scale power systems is highly dependent on the observability of the system, as provided by the selected buses for local measurements of the system frequency.

The existing contingency-based methods for inertia monitoring have advantages, however, they present certain disadvantages, highlighted as: 1) these methods do not account for the effects of virtual inertia provided by RESs. Virtual inertia provided by RESs can significantly influence the total equivalent system inertia. Ignoring this aspect may lead to incomplete assessments of the power system's inertia status. 2) the aforementioned methods do not incorporate measurement noise into the dataset used for training and validating the ML-based method. For practical scenarios, measurement noise is a common concern in the



Fig. 11. Typical process of system inertia monitoring based on ML-based method.

power system's data collection. Ignoring the measurement noise in the dataset may affect the accuracy and robustness of inertia monitoring. Table 2 summarizes the properties and relative comparisons of contingency operation-based methods.

4.2. Normal Operation-Based methods

Continuously updating the status of system inertia in real- or nearreal-time is essential for power system operators to ensure grid stability and manage the dynamics of modern power systems. Monitoring system inertia during normal operation, as opposed to relying solely on contingency-based methods, provides a valuable alternative. The primary objective of these approaches is to monitor system inertia during normal operation using ML-based methods. The data for constructing the ML-based method can be obtained from the historical data of the actual power systems or simulation models of the power system. Other useful data sources can also be incorporated to enhance the accuracy of the ML-based method.

For accurate monitoring the time-varying inertia using ML-based methods, two conditions should be met: 1) The implemented ML model should be capable of processing temporal data; and 2) The training dataset, including input features and output inertia, must adequately represent the time-varying operational conditions of the power system. For example, inertia data, which serves as label information, can be associated with selected input features measured in the real power system over a period (e.g., one or two years) to develop the ML-based inertia monitoring method. In this scenario, the time-varying tracking of the ML-based inertia monitoring method depends on the reporting interval of the inertia data. The methods described in [112,113] integrate various data sources, including real-time measurements, data from transmission system operators and the reliability coordinator, and meteorological data, to develop an ML-based method for continuous inertia monitoring. The MRFR method is selected for this purpose due to its capability to handle multiple data sources and provide robust predictions. Frequency measurements from twenty Phasor Measurement Units (PMUs) spread across the U.S. Western Electricity Coordinating Council (WECC) system are collected. The minimum volume enclosing ellipsoid method [114] is then used to construct an ellipsoid from the time-series frequency information, which helps extract informative features that capture the correlation between timevarying system inertia and frequency. Four key parameters derived from the constructed ellipsoid are chosen as descriptive features for inertia monitoring: volume, center vectors, projections of the longest semi-axis along each dimension, and eccentricity. In addition to these graphical parameters, load profiles from transmission system operators and actual weather data from multiple U.S. cities are collected. System inertia data obtained from the reliability coordinator serves as the label for training the ML-based method. The time-varying tracking capability of this ML-based inertia monitoring method depends on the reporting interval of the inertia data provided by the reliability coordinator, which is ten minutes in this case. Given the small reporting interval, the ML-based method can monitor inertia in shorter time windows. This method enables monitoring of the system's available kinetic energy and system inertia on both yearly and daily bases, offering insights into long-term trends and daily variations.

Another approach to harnessing ML-based methods for monitoring system inertia is demonstrated in [115]. This approach employs a multivariate mixture modeling technique that combines non-Gaussian distributions with Markov chains to capture the complex stochastic and specific temporal dependencies between system frequency and inertia. Continuous monitoring of power system inertia is achieved online using only steady-state system frequency as input. Frequency measurements are collected from three Phasor Measurement Units (PMUs) installed at different universities in the United Kingdom, capturing continuous frequency variations over a two-year period. System inertia data, which serves as the target variable for training the ML-based method, is prepared by gathering generation dispatch information from the transmission operator. During the deployment stage, the ML-based method is periodically recalibrated using the generation dispatch information, which enhances the flexibility and accuracy of the monitoring. Similar to the methods proposed in [112,113], the timevarying tracking capability of this ML-based inertia monitoring method depends on the reporting interval of the inertia data provided by the generation dispatch information. An important consideration is the need for a large historical dataset for training, which may pose challenges if such data is not readily available.

The authors in [116] focus on monitoring system-, regional-, and bus-level inertias using an ANN-based method providing inter-area modal information. Frequency measurements are collected from all buses in the IEEE 118-bus system, divided into three regions based on the graph theory [117]. The modal information of the system is determined, including the first four modes of the frequency for each bus. The dataset is created by varying the inertia of SGs and direct-connected motors, which is used for training and testing the ANN-based method. The method provides the monitoring of system-, regional-, and bus-level inertias. However, the results indicate the limited accuracy of

Table 2

Comparison of contingency operation-based inertia monitoring.

Ref.	Type of monitoring	ML model	System type and considerations	Input features	Pros/Cons
[106]	System inertia*	ANN	Two-bus test system without any controller;	Discretized frequency signal	 Simple power system; Local frequency measurement; Ignoring noise in measurements; Ignoring the participation of RESs in virtual inertia;
[107]	System inertia	ResNet	IEEE 39-bus system model; Generator and load events;	f_{COI} and its corresponding RoCoF	 + Considering fco; - Ignoring noise in measurements; - Ignoring the participation of RESs in virtual inertia;
[108]	System inertia	CNN	IEEE 39-bus system model; Generator and load events;	<i>fcor</i> ; disturbance size;	 + Considering fcoi; - Selection of disturbance size as the input feature; - Ignoring noise in measurements; - Ignoring the participation of RESs in virtual inertia;
[109]	System inertia	LSTM	IEEE 39-bus system model; Load events;	Frequency signal; disturbance size;	 Local frequency measurement; Selection of disturbance size as the input feature;
[110]	Regional inertia**	LSTM	IEEE 39-bus system model; Clustering the generators in multiple regions;	Frequency signal of generators; active and reactive power signal of generators and loads; voltage and phase signals of all busses;	 Ignoring the participation of RESs in virtual inertia; Ignoring noise in measurements;
[111]	System inertia em inertia re	ANN fers to th	Single-machine model;	Local measured frequency; RoCoF;	 + computation- efficient method; - Simple power system; - Ignoring the participation of RESs in virtual inertia; - Ignoring noise in measurements; - Challenge in local measurements; (7).
** F	Regional inert	ia refers	to the equivalent i	nertia of a region wi	th the busses which

monitoring bus-level inertia. It is important to note the limitations of this method as this approach is primarily designed for large-scale power systems with SGs and may not be sufficiently suitable for systems with a high penetration of RESs.

In [118], the authors present the development of an inertia monitoring approach based on an MLP-based method. This method utilizes the measurements obtained from the power flow analysis in monitoring system inertia. Input features for the MLP-based method are chosen based on the Pearson correlation coefficient analysis [119]. The selected variables include the total active power generation from all SGs and all converter-interfaced generators, the total power of induction motor loads directly coupled to the system, and the loading of transmission lines. The MLP-based method is trained and validated using the selected input features and the corresponding inertia data obtained from the IEEE 9-bus system model. In the validation stage, the performance of the trained MLP-based inertia monitor is validated by implementing Hardware-In-the-Loop (HIL) testing. One notable limitation of this approach is its validation for a small-scale power system.

A CNN-based inertia monitoring method that focuses on extracting area-level inertia using the voltage signals from the selected buses is proposed in [120]. In this method, a variable called "momentum" (*M*) is monitored instead of inertia constant by using:

$$M = 2\sum_{i=1}^{N_{SG}} \frac{H_i * S_{r_i}}{f_n}$$
(12)

The reason for using momentum for monitoring is that it is an incremental variable; when a component with inertia is connected to the grid, the momentum increases. Unlike the inertia constant, which may remain unchanged if the inertia of the new component matches the center of inertia (the average weighted inertia of the system), momentum effectively captures changes in inertia as components are connected or disconnected. To validate this approach, synthetic data is generated by simulating the IEEE 39-bus benchmark system, which is divided into four areas [121]. The test system is modified to model variations in load power more realistically. Time-varying voltage measurements from a limited number of buses are selected, and their harmonics are used as input features for the CNN-based method to monitor momentum in the selected area. By varying the inertia of SGs in each area of the test system and measuring the voltages at the selected buses, a training dataset is prepared to establish the ML-based inertia monitoring method. This developed method can continuously monitor the time-varying system inertia.

The study described in [122] presents a method for monitoring the equivalent system inertia, which includes the inertia of SGs and the virtual inertia supported by the virtual synchronous generator (VSG) operation of RESs at the distribution level. Additionally, it considers the damping effects of RESs. The study takes into account various RESs, including WTG and PV farms, and BESSs equipped with the VSG control. These sources have technical limitations, such as dead band, delay, and saturation, which impact their behavior and interaction with the grid. The VSG block diagram is illustrated in Fig. 12, highlighting the nonlinear properties. H_{VSG} and D_{VSG} are the virtual inertia and damping of each source, respectively. PINN based on RNN is proposed to monitor the equivalent system inertia while considering the VSG control of RESs. The monitoring includes the inertia of SGs, the virtual inertia provided by VSG operation in RESs (H_{VSG}), and the damping effect (D_{VSG}). The validation process involves the collection of data from the European medium voltage distribution system with stochastic load variations, enhancing its applicability to practical scenarios. Input features for the PINN-based method encompass the frequency signals of SG, the active powers of SG, WTG farms, and BESSs. It also seeks to establish connections between the weather-related data and virtual inertia of RESs, recognizing the role of environmental conditions in grid behavior.

The research discussed in [123] addresses the monitoring of regional inertia by utilizing the frequency and power data from PMUs. This method determines the inertia constant for each region based on the



Fig. 12. VSG block diagram with nonlinear properties.

Table 3

Comparison of normal operation-based inertia monitoring.

Ref.	Type of monitoring	ML	System type and	Input	Pros/Cons
[112,113]	Yearly and daily system inertia	MRFR	U.S. WECC system	Four constructed ellipsoid parameters; inertia data; load profile; weather data;	 + Monitoring system inertia in the real power system; - Availability of real-time inertia data and load profile are challengeinger
[115]	Time-varying system inertia	Markov chain	U.K. power system	Frequency signals	 Honitoring system inertia in the real power system; Requiring a large historical data:
[116]	System inertia; Regional inertia; Bus inertia;	ANN	IEEE 118-bus system model; Consideration of Inertia of generators and directly- connected motors;	Modal information of frequency signals;	 + Monitoring of three levels of inertia; - Limited accuracy especially at bus level; - Monitoring of inertia for power system with only SGs:
[118]	System inertia	ANN	IEEE 9-bus system model; SGs are replaced by WPP; Dynamic loads are considered;	Active power from all SGs; Active power produced by all CIGs; Total induction motor loads;	 + Verification of method by the experiments; - Small power system;
[120]	Area-level momentum	CNN	IEEE 39-bus system model;	Harmonics of bus voltages;	 Determining the areas and busses for measurement is challenging; Ignoring noise in measurements:
[122]	Equivalent inertia and damping coefficient of VSG operation of RESs	PINN	European medium voltage distribution system; Stochastic variations of load; VSG operation of RESs associated with popliaeaction	Frequency signal of SG; Active powers of SG, WTG farm, and BESS; Irradiation and wind speed data;	 + Monitoring of Equivalent inertia including the inertia of SGs and virtual inertia of SGs; + Monitoring of VSG parameter of RESs; - Ignoring noise in measurements
[123]	Regional available kinetic energy	RL	IEEE 39-bus system model;	Frequency and power of all generator's busses;	H measurements; + Do not need any offline training; - Ignoring the penetrations of RESs and their frequency contributions;

electrical power and kinetic energy variations, assuming constant mechanical power within the frequency extremums during normal operations. To address the inaccuracies resulting from the variations in the mechanical power of SGs, an RL algorithm is employed to detect the changes in the mechanical power, thus eliminating invalid inertia values. It is worth noting that the main limitation of this method is its poor performance for the power systems with the high penetration of RESs.

It is a point of concern that system inertia monitoring during normal operation may not provide accurate information for different contingencies, where system inertia can differ from its steady-state value. It is exemplified when transmission line or transformer outages occur and power system is separated into several areas with different inertias apart from the steady-state inertia. In this way, for different contingency scenarios, it is essential to determine system inertia accurately so that the dynamic response of the power system during disturbances is correctly estimated. The monitoring plays a critical role in the decision-making process for the system operators, which is essential in maintaining grid stability and undertaking the corrective actions. Additionally, the noise measurement is crucial for the accuracy of inertia monitoring methods. Table 3 summarizes and compares the specifications of normal operation-based methods.

4.3. Perturbation-Based methods

Utilizing the excitation signals generated by the power electronicsinterfaced sources to monitor system inertia is an intriguing approach [124]. It involves perturbing the system in a controlled way without compromising system stability, measuring the resulting output responses, and then using ML-based methods to identify system inertia. This approach can be particularly valuable for monitoring inertia's status under controlled conditions and studying the system's response due to different disturbances.

In [125], a method developed for monitoring system inertia of a single-machine power system consisting of a WTG and a PV system provides valuable insights into inertia monitoring under specific conditions. The size of the perturbation and the initial RoCoF are selected to train and test the ML-based method. Since the input features are static during the same iterative conditions, a feedforward ANN has been selected as model. To calculate the initial RoCoF, the system frequency is measured, and the initial RoCoF is calculated by the derivative of the fifth order polynomial function fitted with measured frequency data [126] as follows:

$$\frac{\Delta f}{f_n} = A^* t^5 + B^* t^4 + C^* t^3 + D^* t^2 + E^* t + F \tag{13}$$

$$InitialRoCoF = \frac{d(\Delta f/f_n)}{dt}\Big|_{t=0}$$
(14)

One notable advantage of this method is its ability to monitor the inertia both during steady-state and disturbance conditions. By including data from both scenarios in the training dataset, the ML-based method can potentially provide a useful inertia monitoring for various operational states. However, the results obtained by this simplified model may not directly translate to large-scale power systems with complex interconnected components. Furthermore, the method does not explicitly consider the measurement noise in the training dataset.

To monitor the time-varying inertia based on perturbation-based methods, the probing signal is continuously and periodically injected into the power system. The effect of this signal injection on the power system, which experiences varying operational conditions in terms of system inertia at different moments, allows for temporal inertia monitoring by measuring the selected variables and feeding them into the ML-based method. Typically, the varying operational conditions of the power system are represented by altering the amplitude of the probing signal and system inertia. An LSTM-based method described in [127] for an islanded hybrid power system with distributed generators, WTG, fuel cells, and BESS presents an interesting approach to continuously monitoring the system inertia in microgrids. The inclusion of PID controllers for RESs to stabilize the system frequency is a practical consideration of this method. A periodic pulse is applied to the system, during the measurement of the system frequency. After adding noise and calculating the RoCoF, the frequency and RoCoF signals are fed to the LSTM model to process and classify temporal information. The training dataset includes frequency and RoCoF signals, with the system inertia collected by varying the system inertia and the amplitude of probing signal in the

International Journal of Electrical Power and Energy Systems 162 (2024) 110279

islanded hybrid power system. The results show that this method yields good performance in terms of the accuracy of time-varying inertia monitoring. Anyhow, the effects of tie-line dynamics have not been considered in this method.

The spatial feature of system inertia is identified by the CNN-based monitoring installed in the BESSs as perturbing and monitoring devices [128,129]. The aggregated single-machine model of the power system proposed in [130] is implemented to collect the dataset, including noisy frequency and RoCoF for training and validating the monitoring method [128]. In [129], the federated CNN-based method is used to monitor the regional inertia in a multi-region power system. In each area, one BESS perturbs the system that is non-synchronized with the other areas, measures the system frequency, and constructs the features to train the CNN-based method, including noisy frequency and corresponding RoCoF signals. After the training, the weights of local CNN models are sent to a central server. The central server updates the weights of each local CNN model based on the weighted average of all reached weights obtained from the clients for each time window. Indeed, the use of the decentralized method for system inertia monitoring can be highly advantageous for scenarios where centralized data collection from multiple areas may not be feasible or practical. This decentralized method does not require collecting a high volume of data from the entire power system, which can alleviate the burden on communication infrastructures and data storage. However, there are some drawbacks regarding the two mentioned methods: 1) the timevarying nature of system inertia is not considered, 2) the simple transfer function model of the power system cannot ensure the feasibility of these methods in real and complex power systems with acceptable accuracy, 3) the role of dynamic loads and RESs is ignored, and 4) the use of local frequency measurement may lead to high errors when nonmonotonic frequency deviation occurs.

To tackle the above-mentioned issues and emulate realistic conditions, system inertia is monitored for a modified IEEE 39-bus test system, considering dynamic loads and WTG plants as shown in [131]. A ramplike probing signal is injected by the converters on both sides of the HVDC transmission line for steady-state and disturbance conditions. The noisy frequency signal and perturbation amplitudes of are measured to train the feedforward ANN model. Several SGs have been replaced by WTG plants, which are not used for frequency control. Static loads based on a ZIP¹ model associated with frequency-dependent parameters are considered for training the ANN model, while the motor loads have been added for testing the model. This method relies on the existing infrastructure comprising HVDC transmission lines, and does not need widespread use of PMUs and extra equipment. However, it is capable of only monitoring the system inertia on two sides of the HVDC transmission line. Moreover, the use of dynamic loads in the only test stage is a questionable comment in terms of monitoring accuracy. Finally, the contribution of the virtual inertia of RESs has not been considered in this method.

The use of ambient measurements is proposed in [132,133]. In [132], the frequency and voltage signals of all buses are measured. Then, the dataset, which encompasses the frequency, corresponding RoCoF, and voltage signals associated with noise is used as the features. The best features, including the frequency and corresponding RoCoF signals within the first second after perturbation, are selected using a wrapper feature selection method. The LRCN-based method is trained and validated using a dataset constructed by varying system inertia and the amplitude of the probing signal in an IEEE 24-bus test system. A method called zero generation injection bus is proposed in [133] to optimize the placement and number of PMUs, and to maximize system's observability as well as to optimize the volume of input features. Besides, the GCN model apart from the LRCN is used for the comparison. Both ML-based methods have been efficient in identifying the important temporal and

graphical information embedded in the collected power system dataset. Due to their inherent limitations in capturing long-term and spatial dependencies within graphs-stemming from the diffusion of information across multiple layers-GCNs may fall short. To address this issue, GATs are utilized in [134] to enhance the accuracy of system inertia monitoring. GATs achieve this by effectively capturing complex relationships and dependencies within power systems, thanks to their ability to assign varying levels of importance to different nodes. The frequency, and corresponding RoCoF signals associated with noise are as the features and the GATs-based method has been trained and tested by using the dataset presented in [132,133]. Table 4 shows the summary and relative comparison of perturbation-based methods.

5. Research and industrial projects related to system inertia monitoring comprising ML-based methods

Table 5 provides an overview of recent and ongoing research projects focused on monitoring system inertia using ML-based methods. In the project reported in [135], an ANN-based method is applied to monitor system inertia during normal system operation. The training dataset includes the cumulative active power of SGs, CIGs, directly connected induction motors, and the power line loading. Furthermore, they conducted HIL testing to validate the practical applicability of their proposed method [118]. Another project, reported in [136], aims to achieve precise day-ahead and intra-day forecasts of kinetic energy as system inertia. In this study, various input features are selected, including demand, generation capacity, interconnected power flow, electricity prices, inertial response, and day-ahead forecasts of loads, transmission systems, and RESs. These features are used to construct a probabilistic ML-based method that accurately and cost-effectively quantifies the forecast uncertainties. The method's performance is evaluated using historical data from the power systems in the United Kingdom and the Nordic region [137,138]. The forecasted inertia could potentially support day-ahead frequency response procurement and real-time system operations. The project described in [139] focuses on perturbationbased inertia monitoring using the LSTM-based method. Specifically, the researchers considered the impact of the frequency contributions of RESs on the frequency of an isolated grid. By measuring the frequency and RoCoF after introducing a perturbation signal, they developed an LSTM-based inertia monitor [127]. Additionally, researchers in [140] explore the effective monitoring of regional inertia and they use MLbased methods to monitor the inertia of regional power systems.

In practical terms, the monitoring of system inertia and kinetic energy in real power systems have gained significant attention. Many power systems have implemented tools for measuring system inertia by monitoring the operational state of SGs. Notably, the Electric Reliability Council of Texas (ERCOT) [141], the Nordic region [142], Vietnam [143], and India [144] have adopted such tools. Then, companies like Reactive Technologies have developed an innovative method to determine the kinetic energy, as illustrated in Fig. 13 [145]. This technology utilizes supercapacitors as modulators to intentionally perturb the power system, inducing fluctuations in the system frequency. The eXtensible Measurement Units (XMUs) distributed throughout the grid detect these frequency variations corresponding to the power pulses and relay the data to the processing platform for monitoring the system inertia.

These ground-breaking devices are initially deployed by the National Grid Electricity System Operator (ESO) in the United Kingdom [146]. The National Grid ESO claimed that United Kingdom customers save \$19.6 million annually on managing system inertia, which is achieved by eliminating the need of purchasing excessive energy reserve margins [147]. Moreover, with precise and real-time measurements of grid inertia, power system operators may enhance the efficiency of mitigating the risk of grid instability and blackouts. These two advantages are illustrated in Fig. 13 for the case of inertia measurement in the United Kingdom [148]. Fig. 14 shows how the potential cost-saving

¹ Impedance(Z), Current (I), Power (P).

Table 4

Comparison of perturbation-based inertia monitoring.

				-	
Ref.	Type of monitoring	ML model	System type and considerations	Input features	Pros/Cons
[125]	System inertia	ANN	Single-machine model associated with a WTG and a PV	Size of perturbation; initial RoCoF;	 + Inertia monitoring during normal and disturbance conditions; - Simple power system model; - Ignoring noise in measurements;
[127]	Microgrid inertia	LSTM	Islanded microgrid including SG, WTG, BESS, and fuel cell; RESs contribute to frequency stabilization;	Noisy frequency signal; Noisy RoCoF signal;	 + Monitoring of time-varying inertia; - Not considering the tie-line dynamics;
[128]	System inertia	CNN	Aggregated single-machine model; BESS perturbs the system;	Noisy frequency signal; Noisy RoCoF signal;	 Simple power system model and measurements; Not considering the dynamic loads and RESs
[129]	Regional inertia	FL	Two-area single- machine model; BESS in each area perturbs the system	Noisy frequency signal; Noisy RoCoF signal;	 + Distributed monitoring of inertia; - Simple power system model and measurements; - Not considering the dynamic loads and RESs
[131]	System inertia	ANN	IEEE 39-bus system model; HVDC Converters perturb the system; SGs are replaced by WTG plants;	Size of perturbation; Noisy frequency signal; Voltage of selected busses;	 + Inertia monitoring during normal and disturbance conditions; + Perturbation and monitoring by the available infrastructure; - Not considering the virtual inertia of RESs; - Only measuring the frequency at one point for monitoring;
[132]	System inertia	LRCN	IEEE 24-bus system model;	Noisy frequency, RoCoF, and voltage signals of all busses;	 Not considering the virtual inertia of RESs;
[133]	System inertia	GCN; LRCN;	IEEE 24-bus system model; Optimal placement of PMUs;	Noisy frequency and RoCoF signals of selected busses; Voltage signals of all buses;	 Not considering the virtual inertia of RESs;
[134]	System inertia	GAT	IEEE 24-bus system model:	Noisy frequency, and RoCoF signals	 Not considering the virtual inertia

(green area) and risk (red area) zones are determined by precisely measuring of the system inertia. Subsequently, this technology has been adopted by numerous power systems worldwide, including Australia [149], Taiwan [150], Japan, New Zealand, and Italy [147]. In Australia, Australian renewable energy agency claimed that continuous measurement of inertia has proven extremely valuable, enhancing grid operations and enabling the integration of greater amounts of solar and wind energy into electricity grids [151]. In addition to these tools and devices used for system inertia monitoring, ERCOT has developed a forecasting tool to determine the kinetic energy over a rolling 168-hour period. This

of all busses;

of RESs:

Table 5

ML-based system inertia monitoring research projects.

Project Name	Founded by (Name/ Country)	Key Features	Test System
Lessons learned from monitoring & forecasting key performance indicators on Impact of power electronics penetration [135]	European Union's horizon 2020 research and innovation programme- European Union	System inertia monitoring; Normal operation- based; ANN-based method; Based on power flow analysis;	IEEE 9-bus system
Short-term system inertia forecast [136]	National Grid ESO through network innovation allowance- United Kingdom	Day ahead system inertia forecast; Quantify forecast uncertainties;	UK and Nordic power systems
Characterization and detection of power system ambient, transient and forced oscillations based on PMU data analytics in Indian context [139]	Central power research institute- India	Perturbation- based; LSTM-based method; Frequency contribution of RESs;	Islanded microgrid model
Effective regional inertia monitoring and automatic control with a whole system approach [140]	Smarter power energy networks transmission- United Kingdom	Regional inertia monitoring;	

forecast relies on the unit commitment plans submitted by the generators on an hourly basis [152]. Reliability unit commitment is necessary during periods when inertia levels are insufficient. ERCOT also incorporates historical system inertia data to assess the requirements for responsive reserve service, adjusting the amounts needed based on varying inertia conditions [153].

Despite the existence of established monitoring and forecasting tools and devices, for many power systems with high level of RES integration, companies specializing in power system control and operation have begun to develop inertia monitoring tools and devices based on MLbased techniques. One noteworthy example is General Electric (GE), which has introduced a system inertia monitor utilizing ML-based methods [143]. This innovative tool is designed to measure and predict effective inertia on both regional- and global-scale. It claims to provide a comprehensive measure of effective inertia, encompassing the inertia of the rotating synchronous generations as well as the inertia-like effects of non-synchronous generations, such as WTG and PV systems, as well as the passive responses from domestic and industrial demand. By applying ML-based methods, inertia can be forecast, utilizing the data from both generation and demand. Regional inertia is also monitored using the area RoCoF and net boundary power, derived from the measurements of grid frequency and power flows. Recently, this device has been implemented in the National Grid ESO of the United Kingdom [154]. Siemens Energy is another company introducing a system for monitoring system inertia through power electronics converters. These converters can be periodically activated to provide crucial information to system operators. The method developed by Siemens Energy involves injecting a known perturbation by using converters and analyzing the network's frequency response to this perturbation to monitor the system inertia. This monitoring technology employs an ML-based method to accurately monitor system inertia by taking into account real power system features such as generator controller actions, load-frequency dependencies, and power system oscillation modes [155].



Fig. 13. Architecture of system inertia monitor developed by [145].



Fig. 14. Result of real-time monitoring of system inertia in the United Kingdom [148].

6. Applications of system inertia monitoring and ML-based methods for power system studies

Monitoring spatial-temporal system inertia in modern power systems offers opportunities for its application in diverse areas such as analysis, control, protection, and optimization. Moreover, ML-based methods can leverage this potential to tackle the challenges posed by complex power systems that are heavily integrated with RESs, thanks to their powerful capability to handle large amounts of multi-dimensional data.

The estimation of frequency response characteristics stands out as one of the most intriguing applications of system inertia monitoring in conjunction with ML-based methods. Predicting the frequency nadir has become a more challenging task due to the increasing penetration of RESs, which have significantly altered the generation portfolio of the power systems. In [156,157], the frequency nadir is estimated by employing ANN as the model. In [156], the input features of the ANN include the disturbance size and available kinetic energy of the system as system inertia. Meanwhile, [157] uses the load bus voltages, frequency variations after an event, system inertia, and event size to train the ANNbased method. To determine the best model and dataset in terms of estimating accuracy of frequency nadir, a comparative analysis is conducted in [158], where five types of models and two datasets are evaluated. The applied models include ANN, XGBoost, linear regression, gradient boosting, and support vector regression. The datasets used for training encompass the unit generation dataset, and system inertia and headroom dataset. Ultimately, XGBoost- and gradient boosting-based methods trained by the system inertia and headroom datasets achieve the highest estimating accuracy. In another research effort [159], the focus shifts to estimate the frequency response parameters following a disturbance, including frequency nadir, maximum RoCoF, and steadystate frequency. XGBoost-based algorithm is employed for prediction and is trained using variables such as system inertia, total system load

demand, system spinning reserve, aggregated available kinetic energy from online SGs, disturbance size, and event location.

Power system operatorsrely on predicting frequency response characteristics to assess the frequency security of the system, as these characteristics are indicative of the impact of a sudden loss of a generating resource or connection of a large load. The monitoring of frequency response characteristics holds significance in applications such as frequency-constrained unit commitment and economic dispatch. In the context of security-constrained unit commitment and economic dispatch, the prediction of frequency nadir and RoCoF is addressed using ANN as a supervised learning-based approach reported in [160]. This ANN-based method takes into account the key features, including the power outputs of thermal generation units, and WTG and PV plants, spinning reserves, and system inertia. This method not only determines the additional inertia required for the islanded power systems with high RESs penetration but also aims to minimize the associated costs related to the inertia support from the energy sources. In [161], the estimation of frequency nadir and a parameter called stepwise frequency constraint, which enhances frequency recovery performance, are explored for the frequency-constrained unit commitment. Here, system inertia is among the input features used to train and validate a conservative sparse neural network-based algorithm. The output of the MLbased method is then transformed into constraints that are incorporated in conventional unit commitment model.

The objective of the ML-based method proposed in [162] is to enhance the accuracy of disturbance size estimation compared to the traditional estimation methods. In this approach, the disturbance size estimations generated by three traditional methods, based on equation (10), utilizing parameters such as system inertia and RoCoF (including locally measured RoCoF, the median of all measured frequencies, and the center of inertia), serve as the inputs to an MLP-based method. This method is trained by data collected from thirty-two confirmed disturbances in the WECC system to minimize the estimation errors associated with the traditional methods.

Determining the fast frequency reserve is yet another valuable application of system inertia monitoring in combination with ML-based methods. In [163], the frequency nadir estimation plays a pivotal role in a planning process that quantifies the capacity of fast-responsive reserve resources in the ancillary service market. This proposed plan has proven successful in maintaining the frequency nadir above the security criterion and preventing under-frequency load shedding across the various penetration levels of RESs. An XGBoost-based method is trained and deployed to predict the frequency nadir, with system inertia being one of the selected features. In [112,164], the target is to procuring the fast frequency reserve for PV plants participating in the system frequency stability. Here, an ANN-based method is trained using system inertia, system governor capacity, and frequency nadir set point. ANN-based method is employed to adjust the headroom set points of the frequency controllers for the PV plants. The results indicate headroom savings of at least 40 % compared to the traditional strategies, showcasing the effectiveness of this approach in optimizing frequency reserve management. A fast security monitoring method has been proposed by integrating ML-based methods in [165]. This method employs the Gaussian processing regression model to monitor the security of the power system. Additionally, the incremental Naïve Bayes algorithm is

used to estimate the amount of load to be shed and determine the specific bus for the load shedding operations. The ML-based methods are trained using various parameters, including system inertia, load levels, fault locations, and power dispatched from RESs. Finally, in [166] a methodology is proposed to determine an optimal load shedding amount to prevent enabling the under/over-frequency protective relays based on ML-based methods. The target of this approach is to determine the amount of load to be increased and decreased in islanded areas following a separation of power system in two islands due to tie-line tripping. An ANN-based method is trained and evaluated by the features encompassing total generation and load, system inertia, power output and inertia of each generator, and power of each load.

While ML-based applications utilizing system inertia have been introduced in the literature, there remain numerous unexplored avenues for leveraging ML-based methods in conjunction with system inertia monitoring. Moreover, the existing applications have the potential to improve and further develop existing limitations. Table 6 summarizes the applications of system inertia monitoring and ML-based methods.

7. Conclusion and future directions

This paper provides a comprehensive overview of existing inertia monitoring methods that utilize ML-based techniques, offering guidance for researchers involved in the application of intelligent methods for inertia monitoring as well as discussing academic and industrial projects related to this subject. We have examined the ML-based methods employed in inertia monitoring to shed light on how knowledge is extracted based on the type of training features used. The monitoring methods have been categorized into three main groups: contingency-, normal operation-, and perturbation-based methods, and we have discussed the advantages and disadvantages of each group. Although there is a limited amount of literature regarding the applications of monitored inertia in fields utilizing ML-based methods, we have explored these applications within the scope of this review. We anticipate that this review will serve as a valuable reference and source of inspiration for future research in this area. Despite the significant progress made, there

Table 6

Applications of system inertia monitoring and ML-based methods.

Application	Target	ML model	Ref.
Frequency response characteristics	Estimation of frequency nadir	ANN XGBoost and gradient boosting	[156,157] [158]
estimation	Estimation of frequency nadir, max. RoCoF, steady-state frequency	XGBoost	[159]
Security- constrained unit commitment	Optimization of cost of inertia support of RESs by estimating frequency nadir and RoCoF	ANN	[160]
	Estimation of frequency nadir and stepwise frequency constraint	Conservative sparse neural network	[161]
Disturbance size estimation	Improve the accuracy of traditional estimation methods	MLP	[162]
Determining fast frequency reserve	Determination of fast- responsive reserve to avoid load shedding	XGBoost	[163]
	Adjust headroom set point of frequency controller of PV plants	ANN	[112,164]
Security monitoring and load shedding plan	Security prediction and determination of the location and amount of load to be shed	Gaussian processing regression and incremental Naïve	[165]
	Determination of the optimal load shedding amount	Bayes ANN	[166]

remain several unresolved issues, some of which are highlighted below:

- Monitoring of the virtual inertia of RESs: The interplay between the virtual inertia provided by RESs and the spatial-temporal distribution of system inertia is of paramount importance. On one side, RESs contribute to complex virtual inertia that depends on factors such as synchronous inertia (inertia from online SGs), frequency dynamics, and their operational constraints. On the other hand, synchronous inertia varies based on the number of online synchronous generators, influenced by the amount of virtual inertia supported by RESs and weather conditions. This complex interdependence between synchronous inertia, virtual inertia, and frequency dynamics must be taken into account when monitoring system inertia. Given that ML-based methods excel at extracting knowledge from complex and voluminous datasets with high dependencies, they can be effectively employed to monitor both virtual inertia and system inertia. It is worth noting that existing inertia monitoring methods using ML-based algorithms often overlook this intricate dependency and do not consider the monitoring of virtual inertia. However, there is a limited amount of literature on monitoring the equivalent virtual inertia of RESs using ML-based methods, particularly when considering the frequency and operational limitations of these energy sources [167]. Therefore, a promising research direction involves developing ML-based approaches to monitor system inertia, particularly in conjunction with the virtual inertia supported by RESs, including WTG and PV plants, while accounting for their operational constraints in both normal and contingency operations. This approach would provide a more comprehensive and accurate assessment of system inertia in modern power systems.
- System inertia monitoring in the case of inaccurate or unavailable inertia information: In all aforementioned inertia monitoring methods, the inertia data is available and is used in the training data. The ML-based problem is supervised learning. In many real-world scenarios, obtaining precise inertia information can be challenging, and this limitation must be considered when developing ML-based techniques for inertia monitoring. Besides, the development of methods for enriching the available datasets can make ML-based methods to be more accurate and robust.
- Cyber-security considerations: In the existing studies, ML-based algorithms have shown remarkable capabilities in terms of accuracy and performance for a wide range of applications. However, MLbased methods are indeed heavily influenced by the quality and trustworthiness of the data they operate on. As power systems become increasingly connected and reliant on data from various sources, they indeed become more susceptible to threats like false data injection and denial of service attacks, which can significantly impact the stability and reliability of the grid. In this regard, it is critically important to assess the vulnerability and potential impact of ML-based monitoring methods, particularly in the context of the growing threat of cyberattacks. The development of a methodological framework, as outlined in [168], to assess the vulnerability of ML-based inertia monitoring methods against data integrity attacks is a commendable step toward enhancing the security and reliability of power systems. There still exists a need for further impactful research and development to assess the vulnerability of ML-based methods in terms of cybersecurity, detect the attacks, and build robust ML-based system inertia monitoring methods for mitigating cyberattacks by conducting strategies such as adversarial training, anomaly detection methods, and so on.
- **HIL testing:** HIL methodology plays a pivotal role in advancing and testing complex systems across various industrial applications. It achieves this by seamlessly integrating a real-time simulation platform with the hardware and control devices under examination. The inclusion of power and flexibility offered by the HIL framework positions this testing method as a highly promising tool for versatile

International Journal of Electrical Power and Energy Systems 162 (2024) 110279

testing and validation of applicable methods in low inertia power systems, both at the equipment and system-wide levels [169,170]. In the current landscape of literature concerning system inertia monitoring by ML-based methods, most developed methods are constructed and validated within simulation studies, despite using HIL testing in inertia monitoring based on other methods [171]. However, compared to conventional simulation tools, HIL simulation offers a more efficient, rapid, and realistic real-time simulation with heightened accuracy. Numerous technologies including OPAL-RT [172–176], Real-Time Digital Simulator (RTDS) [177,178], National Instruments (NI) [179,180], Speedgoat [181], and Programmable Logic Controller (PLC) [182], have been developing their capabilities to accommodate ML-based approaches. This makes it an attractive avenue for the development and exploration of ML-based system inertia monitoring methods. It is a possible future research direction with the potential to address various limitations and challenges in this field, including real-time interaction between the ML-based algorithms and physical hardware, and hardware compatibility and adaptability. Further future direction can be seen in scalability and complexity considerations in large-scale power systems or complex ML-based methods, cost and availability of the necessary system hardware, and safety and risk factors associated with HIL testing.

- Applications of system inertia monitoring and ML-based methods: The significant potential of applications that leverage both system inertia monitoring and ML-based methods can enhance the analysis and decision-making processes of modern power systems. These applications span various aspects of power system operation and management and can greatly benefit from the synergy between system inertia information and ML capabilities. The applications can include frequency security and stability assessments, frequency control of RESs, load shedding plans, design of protection plans, reserve planning, inertia allocation and market, inertia-based optimal power flow, and unit commitment. While some of these applications have been reviewed and explored in the current paper and related research, there is still substantial room for further investigation and development in the field of power system studies.
- Application of near real-time ML-based methods: As known, the quality of training procedures is determining the efficacy of MLbased techniques for different applications in electrical power and energy systems. In this context, ML-based approaches for inertia monitoring can underperform when the system state is not close to the system state used for training purposes. One of the typical challenges can be frequent changes in system topology, which can be caused by intentional, or/and unintentional topological changes. In conclusion, the offline training sets may not be sufficient enough to represent adequately the actual system behavior. Therefore, by using (near) real-time training data sets and updating the ML-based algorithm, the quality of the system inertia could be improved. The described approach is referred to a class of the incremental learning methods. In [183] such an approach was successfully applied. We trust that the incremental learning-based approach has a good potential for improving the efficacy of inertia monitoring algorithm and it should be further explored in the future.

CRediT authorship contribution statement

Mahdi Heidari: Writing – original draft, Investigation, Conceptualization. Lei Ding: Supervision, Investigation, Conceptualization. Mostafa Kheshti: Writing – review & editing, Conceptualization. Weiyu Bao: Writing – review & editing, Project administration, Conceptualization. Xiaowei Zhao: Resources. Marjan Popov: Writing – review & editing, Conceptualization. Vladimir Terzija: Writing – review & editing, Supervision, Conceptualization.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

No data was used for the research described in the article.

Acknowledgements

This work was supported by National Science Foundation of China (U22B20101). M. Kheshti and X. Zhao are funded by the UK Engineering and Physical Sciences Research Council (EPSRC) under grant EP/T021713/1.

References

- Petroleum B. Statistical Review of World Energy 2022, https://www.bp.com/co ntent/dam/bp/business-sites/en/global/corporate/pdfs/energy-economi cs/statistical-review/bp-stats-review-2022-full-report.pdf; 2022 [Accessed 15 August 2024].
- [2] Ulbig A, Borsche TS, Andersson G. Impact of low rotational inertia on power system stability and operation. IFAC Proceedings Volumes 2014;47:7290–7.
- [3] Hatziargyriou N, Milanović J, Rahmann C, Ajjarapu V, Cañizares C, Erlich I, et al. Stability definitions and characterization of dynamic behavior in systems with high penetration of power electronic interfaced technologies. IEEE PES Technical Report PES-TR77 2020.
- [4] National Grid ESO. Technical report on the events of 9 august 2019, https:// www.nationalgrideso.com/document/152346/download/; 2019 [Accessed 15 August 2024].
- [5] Wu X, Lan Q, Li Z, Zhuang K, Miao Y, Li W. Frequency characteristics of East China Power Grid after bipolar locking of ultra-high voltage direct current (UHVDC). The Journal of Engineering 2017;2017:1237–41.
- [6] Operator AEM. Preliminary report: black system event in south australia on 28 september 2016, https://www.aemo.com.au/-/media/files/electricity/nem/mar ket_notices_and_events/power_system_incident_reports/2017/integrated-final-re port-sa-black-system-28-september-2016.pdf; 2016 [Accessed 15 August 2024].
- [7] Suresh V, Sutradhar R, Mandal S, Debnath K. Near miss of blackout in southern part of north eastern grid of India. 2020 3rd International Conference on Energy, Power and Environment: Towards Clean Energy Technologies: IEEE; 2021. p. 1-5.
- [8] European Network of Transmission System Operators for Electricity (ENTSO-E). Continental Europe Synchronous Area Separation on 08 January 2021, ICS Investigation Expert Panel - Final Report, https://eepublicdownloads.entsoe. eu/clean-documents/SOC%20documents/SOC%20Reports/Continental%20 Europe%20Synchronous%20Area%20Separation%20on%2008%20January% 202021%20-%20Main%20Report_updated.pdf; 2021 [Accessed 15 August 2024].
- [9] Ratnam KS, Palanisamy K, Yang G. Future low-inertia power systems: Requirements, issues, and solutions-A review. Renew Sustain Energy Rev 2020; 124:109773.
- [10] Yang D, Wang X, Chen W, Yan G, Jin Z, Jin E, et al. Adaptive Frequency Droop Feedback Control-Based Power Tracking Operation of a DFIG for Temporary Frequency Regulation. IEEE Trans Power Syst 2024;39:2682–92.
- [11] Wilson D, Yu J, Al-Ashwal N, Heimisson B, Terzija V. Measuring effective area inertia to determine fast-acting frequency response requirements. Int J Electr Power Energy Syst 2019;113:1–8.
- [12] Tan B, Zhao J, Netto M, Krishnan V, Terzija V, Zhang Y. Power system inertia estimation: Review of methods and the impacts of converter-interfaced generations. Int J Electr Power Energy Syst 2022;134:107362.
- [13] Sarker IH. Machine learning: Algorithms, real-world applications and research directions. SN computer science 2021;2:160.
- [14] Abu-Rub OH, Fard AY, Umar MF, Hosseinzadehtaher M, Shadmands MB. Towards intelligent power electronics-dominated grid via machine learning techniques. IEEE Power Electron Mag 2021;8:28–38.
- [15] Duchesne L, Karangelos E, Wehenkel L. Recent developments in machine learning for energy systems reliability management. Proc IEEE 2020;108:1656–76.
- [16] Chen X, Jang Y, Terzija V, Lu C. Review on measurement-based frequency dynamics monitoring and analyzing in renewable energy dominated power systems. Int J Electr Power Energy Syst 2024;155:109520.
- [17] Zhang Y, Shi X, Zhang H, Cao Y, Terzija V. Review on deep learning applications in frequency analysis and control of modern power system. Int J Electr Power Energy Syst 2022;136:107744.
- [18] Wang J, Pinson P, Chatzivasileiadis S, Panteli M, Strbac G, Terzija V. On machine learning-based techniques for future sustainable and resilient energy systems. IEEE Trans Sustainable Energy 2022;14:1230–43.
- [19] Dimoulias SC, Kontis EO, Papagiannis GK. Inertia estimation of synchronous devices: Review of available techniques and comparative assessment of conventional measurement-based approaches. Energies 2022;15:7767.

- [20] Hu P, Li Y, Yu Y, Blaabjerg F. Inertia estimation of renewable-energy-dominated power system. Renew Sustain Energy Rev 2023;183:113481.
- [21] Prabhakar K, Jain SK, Padhy PK. Inertia estimation in modern power system: A comprehensive review. Electr Pow Syst Res 2022;211:108222.
- [22] Heylen E, Teng F, Strbac G. Challenges and opportunities of inertia estimation and forecasting in low-inertia power systems. Renew Sustain Energy Rev 2021; 147:111176.
- [23] Tielens P, Van Hertem D. The relevance of inertia in power systems. Renew Sustain Energy Rev 2016;55:999–1009.
- [24] Cummins Generator Technologies. Rotor Inertia, https://www.stamford-avk. com/sites/stamfordavk/files/AGN183_B.pdf; [Accessed 15 August 2024].
 [25] Moebs W. University Physics v1: OpenStax, https://openstax.org/books/universi
- ty-physics-volume-1; 2021 [Accessed 15 August 2024]. [26] Kundur PS, Malik OP. Power system stability and control: McGraw-Hill.
- Education 1994. [27] Fouad A, Anderson P. Power system control and stability. IEEE Series on. Power
- [27] Fould A, Anderson P. Power system control and stability. IEEE Series on. Power Engineering 2003.
- [28] Terzija VV. Adaptive underfrequency load shedding based on the magnitude of the disturbance estimation. IEEE Trans Power Syst 2006;21:1260–6.
- [29] Fernández-Guillamón A, Gómez-Lázaro E, Muljadi E, Molina-García Á. Power systems with high renewable energy sources: A review of inertia and frequency control strategies over time. Renew Sustain Energy Rev 2019;115:109369.
- [30] Murphy KP. Machine learning: a probabilistic perspective. MIT press; 2012.
- [31] Liu C, Rather ZH, Chen Z, Bak CL. An overview of decision tree applied to power systems. International Journal of Smart Grid and Clean Energy 2013;2:413–9.
 [32] Vankayala VSS Rao ND Artificial neural networks and their applications to
- [32] Vankayala VSS, Rao ND. Artificial neural networks and their applications to power systems—a bibliographical survey. Electr Pow Syst Res 1993;28:67–79.
 [33] Haque MT, Kashtiban AM. Application of neural networks in power systems; a
- [36] Indee MT, Rashtolan Jahr, Appleadon of Internal networks in power systems, a review. International Journal of Energy and Power Engineering 2007;1:897–901.
 [34] Wu L, You S, Dong J, Liu Y, Bilke T. Multiple linear regression based disturbance
- [34] Wu L, Fou S, Dong J, Lu F, Bike F. Multiple mean regression based disturbance magnitude estimations for bulk power systems. In: 2018 IEEE Power & Energy Society General Meeting (PESGM): IEEE; 2018. p. 1–5.
- [35] Cai L, Thornhill NF, Kuenzel S, Pal BC. Wide-area monitoring of power systems using principal component analysis and \$ k \$-nearest neighbor analysis. IEEE Trans Power Syst 2018;33:4913–23.
- [36] Zhang J, Jia H, Zhang N. Alternate support vector machine decision trees for power systems rule extractions. IEEE Trans Power Syst 2022;38:980–3.
- [37] Zendehboudi A, Baseer MA, Saidur R. Application of support vector machine models for forecasting solar and wind energy resources: A review. J Clean Prod 2018;199:272–85.
- [38] Tambunan HB, Barus DH, Hartono J, Alam AS, Nugraha DA, Usman HHH. Electrical peak load clustering analysis using K-means algorithm and silhouette coefficient. In: 2020 International Conference on Technology and Policy in Energy and Electric Power (ICT-PEP): IEEE; 2020. p. 258–62.
- [39] Søndergaard HAN, Shaker HR, Jørgensen BN. Energy systems condition monitoring: Dynamic principal component analysis application. In: 2022 IEEE 10th International Conference on Smart Energy Grid Engineering (SEGE): IEEE; 2022. p. 81–7.
- [40] Chi L, Su H, Zio E, Qadrdan M, Li X, Zhang L, et al. Data-driven reliability assessment method of Integrated Energy Systems based on probabilistic deep learning and Gaussian mixture Model-Hidden Markov Model. Renew Energy 2021;174:952–70.
- [41] Zhang Z, Zhang D, Qiu RC. Deep reinforcement learning for power system applications: An overview. CSEE J Power Energy Syst 2019;6:213–25.
- [42] Chen X, Qu G, Tang Y, Low S, Li N. Reinforcement learning for selective key applications in power systems: Recent advances and future challenges. IEEE Trans Smart Grid 2022;13:2935–58.
- [43] Mnih V, Kavukcuoglu K, Silver D, Rusu AA, Veness J, Bellemare MG, et al. Human-level control through deep reinforcement learning. Nature 2015;518: 529–33.
- [44] Dong H, Zhang J, Zhao X. Intelligent wind farm control via deep reinforcement learning and high-fidelity simulations. Appl Energy 2021;292:116928.
- [45] Dong H, Xie J, Zhao X. Wind farm control technologies: from classical control to reinforcement learning. Progress in Energy 2022;4:032006.
- [46] Xie J, Dong H, Zhao X. Power Regulation and Load Mitigation of Floating Wind Turbines via Reinforcement Learning. IEEE Trans Autom Sci Eng 2023.
- [47] Dong H, Zhao X. Wind-farm power tracking via preview-based robust reinforcement learning. IEEE Trans Ind Inf 2021;18:1706–15.
- [48] Xu H, Sun H, Nikovski D, Kitamura S, Mori K, Hashimoto H. Deep reinforcement learning for joint bidding and pricing of load serving entity. IEEE Trans Smart Grid 2019;10:6366–75.
- [49] Liang Y, Zhao X, Sun L. A Multiagent Reinforcement Learning Approach for Wind Farm Frequency Control. IEEE Trans Ind Inf 2022;19:1725–34.
- [50] Bahrami S, Wong VW, Huang J. An online learning algorithm for demand response in smart grid. IEEE Trans Smart Grid 2017;9:4712–25.
- [51] Duan J, Shi D, Diao R, Li H, Wang Z, Zhang B, et al. Deep-reinforcement-learningbased autonomous voltage control for power grid operations. IEEE Trans Power Syst 2019;35:814–7.
- [52] Xu T, Overbye T. Real-time event detection and feature extraction using PMU measurement data. 2015 IEEE International Conference on Smart Grid Communications (SmartGridComm): IEEE; 2015. p. 265-70.
- [53] Abedinia O, Amjady N, Zareipour H. A new feature selection technique for load and price forecast of electrical power systems. IEEE Trans Power Syst 2016;32: 62–74.

- [54] Jiang Q, Zhu L, Shu C, Sekar V. Multilayer perceptron neural network activated by adaptive Gaussian radial basis function and its application to predict lid-driven cavity flow. Acta Mech Sin 2021;1–16.
- [55] Kadakadiyavar S, Prasad AN, Pareek PK, Vani V, Rekha V, Nirmala G. Recognition efficiency enhancement of control chart pattern using ensemble MLP neural network. Int J Health Sci 2022;4295–306.
- [56] Sabih M, Umer M, Farooq U, Gu J, Balas MM, Asad MU, et al. Image processing based fault classification in power systems with classical and intelligent techniques. J Intell Fuzzy Syst 2022;43:1921–32.
- [57] Wu H, Zhou H, Wang A, Iwahori Y. Precise Crop Classification of Hyperspectral Images Using Multi-Branch Feature Fusion and Dilation-Based MLP. Remote Sens (Basel) 2022;14:2713.
- [58] Rajabi S, Roozkhosh P, Farimani NM. MLP-based Learnable Window Size for Bitcoin price prediction. Appl Soft Comput 2022;129:109584.
- [59] Sarkar S, Kumar S, Roy A, Das B. Multilayer Perceptron Based Early On-Site Estimation of PGA During an Earthquake. In: ICDSMLA 2021: Proceedings of the 3rd International Conference on Data Science, Machine Learning and Applications: Springer; 2023. p. 313–26.
- [60] Shadi MR, Ameli M-T, Azad S. A real-time hierarchical framework for fault detection, classification, and location in power systems using PMUs data and deep learning. Int J Electr Power Energy Syst 2022;134:107399.
- [61] Zhao L, Zhang X, Peng X. Power fluctuation mitigation strategy for microgrids based on an LSTM-based power forecasting method. Appl Soft Comput 2022;127: 109370.
- [62] Samant RM, Bachute MR, Gite S, Kotecha K. Framework for deep learning-based language models using multi-task learning in natural language understanding: A systematic literature review and future directions. IEEE Access 2022;10: 17078–97.
- [63] Alrumiah SS, Alshargabi AA. A Deep Diacritics-Based Recognition Model for Arabic Speech: Quranic Verses as Case Study. IEEE Access 2023.
- [64] Yonglan L, Wenjia H. English-Chinese machine translation model based on bidirectional neural network with attention mechanism. Journal of Sensors 2022; 2022.
- [65] Li Z, Liu F, Yang W, Peng S, Zhou J. A survey of convolutional neural networks: analysis, applications, and prospects. IEEE transactions on neural networks and learning systems. 2021.
- [66] Kattenborn T, Leitloff J, Schiefer F, Hinz S. Review on Convolutional Neural Networks (CNN) in vegetation remote sensing. ISPRS J Photogramm Remote Sens 2021;173:24–49.
- [67] Fan X, Feng X, Dong Y, Hou H. COVID-19 CT image recognition algorithm based on transformer and CNN. Displays 2022;72:102150.
- [68] Dharma EM, Gaol FL, Warnars H, Soewito B. The accuracy comparison among word2vec, glove, and fasttext towards convolution neural network (cnn) text classification. J Theor Appl Inf Technol 2022;100:31.
- [69] Almeida JS, Huang C, Nogueira FG, Bhatia S, de Albuquerque VHC. EdgeFireSmoke: A Novel Lightweight CNN Model for Real-Time Video Fire-Smoke Detection. IEEE Trans Ind Inf 2022;18:7889–98.
- [70] Copiaco A, Himeur Y, Amira A, Mansoor W, Fadli F, Atalla S, et al. An innovative deep anomaly detection of building energy consumption using energy time-series images. Eng Appl Artif Intel 2023;119:105775.
- [71] Nandhini K, Thailambal G. CNN-OSBO Encoder-Decoder Architecture for Drug-Target Interaction (DTI) Prediction of Covid-19 Targets. In: 2022 6th International Conference on Electronics, Communication and Aerospace Technology: IEEE; 2022. p. 1199–206.
- [72] Brintha N, Nagaraj P, Tejasri A, Durga BV, Teja MT, Kumar MNVP. A food recommendation system for predictive diabetic patients using ANN and CNN. In: 2022 7th International Conference on Communication and Electronics Systems (ICCES): IEEE; 2022. p. 1364–71.
- [73] Zhang S, Tong H, Xu J, Maciejewski R. Graph convolutional networks: a comprehensive review. Computational Social Networks 2019;6:1–23.
- [74] Deng L, Lian D, Wu C, Chen E. Graph Convolution Network based Recommender Systems: Learning Guarantee and Item Mixture Powered Strategy. Adv Neural Inf Proces Syst 2022;35:3900–12.
- [75] Qiao P, Zhang Z, Li Z, Zhang Y, Bian K, Li Y, et al. TAG: Joint Triple-hierarchical Attention and GCN for Review-based Social Recommender System. IEEE Trans Knowl Data Eng 2022.
- [76] Chen BY, Ma Y, Wang J, Jia T, Liu X, Lam WH. Graph convolutional networks with learnable spatial weightings for traffic forecasting applications. Transportmetrica A: Transport Science 2023:1–30.
- [77] Hossain RR, Huang Q, Huang R. Graph convolutional network-based topology embedded deep reinforcement learning for voltage stability control. IEEE Trans Power Syst 2021;36:4848–51.
- [78] Liao W, Bak-Jensen B, Pillai JR, Wang Y, Wang Y. A review of graph neural networks and their applications in power systems. J Mod Power Syst Clean Energy 2021;10:345–60.
- [79] Velickovic P, Cucurull G, Casanova A, Romero A, Lio P, Bengio Y. Graph Attention Networks. International Conference on Learning Representations. 2018.
- [80] Zhang K, Sun M, Han TX, Yuan X, Guo L, Liu T. Residual networks of residual networks: Multilevel residual networks. IEEE Trans Circuits Syst Video Technol 2017;28:1303–14.
- [81] Shafiq M, Gu Z. Deep residual learning for image recognition: A survey. Appl Sci 2022;12:8972.
- [82] Yang L, Zhong J, Zhang Y, Bai S, Li G, Yang Y, et al. An improving faster-rcnn with multi-attention resnet for small target detection in intelligent autonomous transport with 6g. IEEE Trans Intell Transp Syst 2022.

M. Heidari et al.

International Journal of Electrical Power and Energy Systems 162 (2024) 110279

- [83] Zhan L, Li W, Min W. FA-ResNet: Feature affine residual network for large-scale point cloud segmentation. Int J Appl Earth Obs Geoinf 2023;118:103259.
- [84] Donahue J, Anne Hendricks L, Guadarrama S, Rohrbach M, Venugopalan S, Saenko K, et al. Long-term recurrent convolutional networks for visual recognition and description. Proceedings of the IEEE conference on computer vision and pattern recognition2015. p. 2625-34.
- [85] Khor H-Q, See J, Phan RCW, Lin W. Enriched long-term recurrent convolutional network for facial micro-expression recognition. In: 2018 13th IEEE international conference on automatic face & gesture recognition (FG 2018): IEEE; 2018. p. 667–74.
- [86] Indumathi N, Divyalakshmi R, Stalin J, Ramachandran V, Rajaram P. Apply Deep Learning-based CNN and LSTM for Visual Image Caption Generator. In: 2023 3rd International Conference on Advance Computing and Innovative Technologies in Engineering (ICACITE): IEEE; 2023. p. 1586–91.
- [87] Raissi M, Perdikaris P, Karniadakis GE. Physics-informed neural networks: A deep learning framework for solving forward and inverse problems involving nonlinear partial differential equations. J Comput Phys 2019;378:686–707.
- [88] Guo Q, Zhao Y, Lu C, Luo J. High-dimensional inverse modeling of hydraulic tomography by physics informed neural network (HT-PINN). J Hydrol 2023;616: 128828.
- [89] Feng D, Tan Z, He Q. Solving River Dynamics at the River-ocean Interface using a Physics-informed Deep Learning Based Data Assimilation Approach. AGU Fall. Meeting Abstracts 2022:H36C-.
- [90] Santoso R, Degen D, Cacace M, Wellmann F. State-of-the-art physics-based machine learning for hydro-mechanical simulation in geothermal applications.
- [91] Cho G, Zhu D, Campbell JJ, Wang M. An LSTM-PINN Hybrid Method to Estimate Lithium-Ion Battery Pack Temperature. IEEE Access 2022;10:100594–604.
- [92] Huang B, Wang J. Applications of physics-informed neural networks in power systems-a review. IEEE Trans Power Syst 2022;38:572–88.
- [93] Misyris GS, Venzke A, Chatzivasileiadis S. Physics-informed neural networks for power systems. In: 2020 IEEE Power & Energy Society General Meeting (PESGM): IEEE; 2020. p. 1–5.
- [94] AbdulRahman S, Tout H, Ould-Slimane H, Mourad A, Talhi C, Guizani M. A survey on federated learning: The journey from centralized to distributed onsite learning and beyond. IEEE Internet Things J 2020;8:5476–97.
- [95] Bonawitz K, Ivanov V, Kreuter B, Marcedone A, McMahan HB, Patel S, et al. Practical secure aggregation for privacy-preserving machine learning. In: proceedings of the 2017 ACM SIGSAC Conference on Computer and Communications Security; 2017. p. 1175–91.
- [96] Li L, Fan Y, Tse M, Lin K-Y. A review of applications in federated learning. Comput Ind Eng 2020;149:106854.
- [97] Cheng X, Li C, Liu X. A review of federated learning in energy systems. IEEE/IAS Industrial and Commercial Power System Asia (I&CPS Asia) 2022;2022:2089–95.
- [98] Sun K, Xiao H, Liu S, Liu Y. Machine learning-based fast frequency response control for a VSC-HVDC system. CSEE J Power Energy Syst 2020;7:688–97.
- [99] Kavitha M, Gnaneswar G, Dinesh R, Sai YR, Suraj RS. Heart disease prediction using hybrid machine learning model. In: 2021 6th international conference on inventive computation technologies (ICICT): IEEE; 2021. p. 1329–33.
- [100] Emmanouil A, Athina M, Carter B, Psounis K. City-wide signal strength maps: Prediction with random forests. Proceedings of the Worldwide Web Conference Association for Computing Machinery. 2019.
- [101] El Mrabet Z, Sugunaraj N, Ranganathan P, Abhyankar S. Random forest regressorbased approach for detecting fault location and duration in power systems. Sensors 2022;22:458.
- [102] Babar B, Luppino LT, Boström T, Anfinsen SN. Random forest regression for improved mapping of solar irradiance at high latitudes. Sol Energy 2020;198: 81–92.
- [103] Sun M, Liu G, Popov M, Terzija V, Azizi S. Underfrequency load shedding using locally estimated RoCoF of the center of inertia. IEEE Trans Power Syst 2021;36: 4212–22.
- [104] Yang D, Li J, Jin Z, Yan G, Wang X, Ding L, et al. Sequential Frequency Regulation Strategy for DFIG and Battery Energy Storage System Considering Artificial Deadbands. Int J Electr Power Energy Syst 2024;155:109503.
- [105] Heidari M, Ding L, Kheshti M. A novel dynamic inertial control of wind turbines based on event size estimation. Int J Electr Power Energy Syst 2024;155:109656.
- [106] Hartono J, Pramana PAA, Kusuma AA, Munir BS. Power System Inertia Estimation Based on Frequency Measurement. In: 2019 Asia Pacific Conference on Research in Industrial and Systems Engineering (APCoRISE): IEEE; 2019, p. 1–4.
- [107] Ramirez-Gonzalez M, Sevilla FS, Korba P. Power System Inertia Estimation Using A Residual Neural Network Based Approach. In: 2022 4th Global Power, Energy and Communication Conference (GPECOM): IEEE; 2022. p. 355–60.
- [108] Tang Z, Hao L, Feng J. Power System Inertia Estimation Based on 1-Dimensional Deep Convolution Neural Network. In: 2022 IEEE 9th International Conference on Power Electronics Systems and Applications (PESA): IEEE; 2022. p. 1–6.
- [109] Tang Z, Hao L, Li J, Chen Q, Li Y, Xu Z. Power System Inertia Estimation Based on Long Short Term Memory Network. In: 2021 IEEE Sustainable Power and Energy Conference (iSPEC): IEEE; 2021. p. 2542–7.
- [110] Shen J, Zhang Y, Ma M, Yang J, Zhao Q, Ke S, et al. Data-driven equivalent inertia partition estimation of power systems. In: 2022 Power System and Green Energy Conference (PSGEC): IEEE; 2022. p. 744–8.
- [111] Das S, Chattopadhyay P. Data-Driven Inertia Estimation of Power Systems using Autoencoder-based Unsupervised Feature Learning. In: 2023 IEEE 3rd Applied Signal Processing Conference (ASPCON): IEEE; 2023. p. 222–6.

- [112] Su Y, Li H, Cui Y, You S, Ma Y, Wang J, et al. An adaptive PV frequency control strategy based on real-time inertia estimation. IEEE Trans Smart Grid 2020;12: 2355–64.
- [113] Cui Y, You S, Liu Y. Ambient synchrophasor measurement based system inertia estimation. In: 2020 IEEE Power & Energy Society General Meeting (PESGM): IEEE; 2020. p. 1–5.
- [114] Ma J, Makarov YV, Diao R, Etingov PV, Dagle JE, De Tuglie E. The characteristic ellipsoid methodology and its application in power systems. IEEE Trans Power Syst 2012;27:2206–14.
- [115] Cao X, Stephen B, Abdulhadi IF, Booth CD, Burt GM. Switching Markov Gaussian models for dynamic power system inertia estimation. IEEE Trans Power Syst 2015;31:3394–403.
- [116] Schmitt A, Lee B. Steady-state inertia estimation using a neural network approach with modal information. In: 2017 IEEE Power & Energy Society General Meeting: IEEE; 2017. p. 1–5.
- [117] Mehrjerdi H, Lefebvre S, Saad M, Asber D. A decentralized control of partitioned power networks for voltage regulation and prevention against disturbance propagation. IEEE Trans Power Syst 2012;28:1461–9.
- [118] Paidi ER, Marzooghi H, Yu J, Terzija V. Development and validation of artificial neural network-based tools for forecasting of power system inertia with wind farms penetration. IEEE Syst J 2020;14:4978–89.
- [119] Zhang Y, Li Y, Song J, Chen X, Lu Y, Wang W. Pearson correlation coefficient of current derivatives based pilot protection scheme for long-distance LCC-HVDC transmission lines. Int J Electr Power Energy Syst 2020;116:105526.
- [120] Linaro D, Bizzarri F, Del Giudice D, Pisani C, Giannuzzi GM, Grillo S, et al. Continuous estimation of power system inertia using convolutional neural networks. Nat Commun 2023;14:4440.
- [121] Ashton PM, Saunders CS, Taylor GA, Carter AM, Bradley ME. Inertia estimation of the GB power system using synchrophasor measurements. IEEE Trans Power Syst 2014;30:701–9.
- [122] Plant R, Babazadeh D, Stock S, Becker C. Real-time inertia estimation in an inverter-dominated distribution grid using a physics-informed recurrent neural network. In: CIRED Porto Workshop 2022: E-mobility and power distribution systems: IET; 2022. p. 940–4.
- [123] Lavanya L, Swarup KS. Continuous Real-Time Estimation of Power System Inertia Using Energy Variations and Q-Learning. IEEE Open Journal of Instrumentation and Measurement 2023;2:1–11.
- [124] Zhang J, Xu H. Online identification of power system equivalent inertia constant. IEEE Trans Ind Electron 2017;64:8098–107.
- [125] Kisinga DA, Makolo P, Trodden P. A Neural Network-based Online Estimation of Stochastic Inertia in a Power System. 2022 IEEE PES/IAS PowerAfrica: IEEE 2022:1–5.
- [126] Inoue T, Taniguchi H, Ikeguchi Y, Yoshida K. Estimation of power system inertia constant and capacity of spinning-reserve support generators using measured frequency transients. IEEE Trans Power Syst 1997;12:136–43.
- [127] Saini P, Parida S. Inertia Estimation of Islanded Power System With Distributed Generation Using Long Short Term Memory Algorithm. In: 2023 IEEE IAS Global Conference on Renewable Energy and Hydrogen Technologies (GlobConHT): IEEE; 2023. p. 1–6.
- [128] Poudyal A, Fourney R, Tonkoski R, Hansen TM, Tamrakar U, Trevizan RD. Convolutional neural network-based inertia estimation using local frequency measurements. In: 2020 52nd North American Power Symposium (NAPS): IEEE; 2021. p. 1–6.
- [129] Poudyal A, Tamrakar U, Trevizan RD, Fourney R, Tonkoski R, Hansen TM. Multiarea inertia estimation using convolutional neural networks and federated learning. IEEE Syst J 2021;16:6401–12.
- [130] Anderson PM, Mirheydar M. A low-order system frequency response model. IEEE Trans Power Syst 1990;5:720–9.
- [131] Pinto RT. Artificial Neural Network-Based Inertia Estimation. In: 2022 IEEE Power & Energy Society General Meeting (PESGM): IEEE; 2022. p. 1–5.
- [132] Tuo M, Li X. Long-Term Recurrent Convolutional Network-based Inertia Estimation using Ambient Measurements. In: 2022 IEEE Industry Applications Society Annual Meeting (IAS): IEEE; 2022. p. 1–6.
- [133] Tuo M, Li X. Machine Learning Assisted Inertia Estimation Using Ambient Measurements. IEEE Trans Ind Appl 2023.
- [134] Albeladi F, Basulaiman K, Barati M. Enhancing System Inertia Estimation: Multi-Head Graph Attention Networks Leveraging PMU Measurements. IEEE Texas Power and Energy Conference): IEEE 2024:1–6.
- [135] Wilson D, Warichet J, Eves M, Al-Ashwal N. 3: Lessons Learned from Monitoring & Forecasting KPIs on Impact of PE Penetration. H2020 MIGRATE project. Tech Rep 2018;D2.
- [136] Ma Z. Short-term System Inertia Forecast. United Kingdom: National grid ESO, https://smarter.energynetworks.org/projects/nia_ngso0020; 2019 [Accessed 15 August 2024].
- [137] Heylen E, Browell J, Teng F. Probabilistic day-ahead inertia forecasting. IEEE Trans Power Syst 2021;37:3738–46.
- [138] Graham J, Heylen E, Bian Y, Teng F. Benchmarking explanatory models for inertia forecasting using public data of the nordic area. In: 2022 17th International Conference on Probabilistic Methods Applied to Power Systems (PMAPS): IEEE; 2022. p. 1–6.
- [139] Parida SK. Characterization and Detection of Power System Ambient, Transient and Forced Oscillations Based on Synchro phasor Data Analytics in Indian Context, India: Central Power Research Institute (CPRI), MoP, Govt. of India, https://www.iitp.ac.in/images/pdf/Advertisement-645_CPRI_234_RA-June-2022 .pdf; 2022 [Accessed 15 August 2024].

M. Heidari et al.

International Journal of Electrical Power and Energy Systems 162 (2024) 110279

- [140] Yu J. Effective Regional Inertia Monitoring and Automatic Control with a Whole System Approach (T2). United Kingdom: Smaterter Power Energy Networks, https://smarter.energynetworks.org/projects/NIA_SPEN_0072; 2022 [Accessed 15 August 2024].
- [141] ERCOT. Inertia: Basic Concepts and Impacts on the ERCOT Grid, https://www. ferc.gov/sites/default/files/2021-11/Inertia_Basic_Concepts_Impacts_On_ERC OT_v0.pdf; 2018 [Accessed 15 August 2024].
- [142] Ørum E, Kuivaniemi M, Laasonen M, Bruseth AI, Jansson EA, Danell A, et al. Future system inertia. ENTSOE, Brussels, Tech Rep. 2015:1-58, https://eepublic downloads.entsoe.eu/clean-documents/Publications/SOC/Nordic/Nordic_rep ort_Future_System_Inertia.pdf; 2015 [Accessed 15 August 2024].
- [143] Global power system transformation consortium. Southeast Asia Community of Practice: Tools and Case Studies for Inertia Monitoring and Calculation, https ://globalpst.org/wp-content/uploads/SE-Asia-COP-Master-Deck_final.pdf; 2019 [Accessed 15 August 2024].
- [144] Kumar A. Assessment of Inertia in Indian Power System. Power System Operation Corporation Limited in collaboration with Indian Institute of Technology Bombay, https://posoco.in/wp-content/uploads/2022/01/Assessment-of-Inertia -in-Indian-Power-System.pdf; 2022 [Accessed 15 August 2024].
- [145] Reactive technologies. Accurate Grid Inertia Measurements, https://reactivetechnologies.com/accurate-grid-inertia-measurements-grid-operators; [Accessed 15 August 2024].
- [146] National grid ESO. Sonar of the power grid: new inertia measurement tools planned for Great Britain's electricity system, https://www.nationalgrideso. com/news/sonar-power-grid-new-inertia-measurement-tools-planned-great-bri tains-electricity-system; 2020 [Accessed 15 August 2024].
- [147] PJM Interconnection. Inertia & System Strength Measurement & Analytics, https ://www2.pjm.com/-/media/committees-groups/forums/emerging-tech/2022/2 0220614/item-4—renewable-integration—inertia-system-strength-measuremen ts-and-analytics.ashx; 2021 [Accessed 15 August 2024].
- [148] Berry B. Inertia estimation methodologies vs measurement methodology: Impact on system operations. CIGRE Symposium Aalborg 2019, https://reactive-tech nologies.com/wp-content/uploads/2022/09/Cigre-Symposium-Aalborg-Paper-1. pdf; 2019 [Accessed 15 August 2024].
- [149] Australian energy market operator. Victorian Inertia Measurement Trial, https ://aemo.com.au/en/initiatives/trials-and-initiatives/victorian-inertia-measure ment-trial/; 2023 [Accessed 15 August 2024].
- [150] Garrett J, Reactive Technologies' Grid Inertia Measurement Tech to Enable Offshore Wind Rollout in Taiwan, financial content, https://markets.financialcont ent.com/stocks/article/bizwire-2023-9-25-reactive-technologies-grid-inertiameasurement-tech-to-enable-offshore-wind-rollout-in-taiwan/; 2023 [Accessed 15 August 2024].
- [151] Australian Renewable Energy Agency, Increasing renewables on the grid through accurate inertia measurement, https://arena.gov.au/news/increasing-renewa bles-on-the-grid-through-accurate-inertia-measurement/; 2023 [Accessed 15 August 2024].
- [152] Du P, Matevosyan J. Forecast system inertia condition and its impact to integrate more renewables. IEEE Trans Smart Grid 2017;9:1531–3.
- [153] North American Electric Reliability Corporation, 2023 Long-Term Reliability Assessment, https://www.nerc.com/pa/RAPA/ra/Reliability%20Assessments% 20DL/NERC_LTRA_2023.pdf; 2023 [Accessed 15 August 2024].
- [154] San Ramos C. National Grid ESO Deploys GE Digital Effective Inertia Metering & Forecasting Solution, General Electric, https://www.ge.com/news/press-releases /national-grid-eso-deploys-ge-digital-effective-inertia-metering-forecasting-sol ution/; 2022 [Accessed 15 August 2024].
- [155] Siemens energy. Grid State Detection Driving the future of electric power transmission by deploying artificial intelligence, https://assets.siemens-energy. com/siemens/assets/api/uuid:e378af9e-86ed-4272-ad5e-9692e64af026/whitep aper-grid-state-detection.pdf; 2021 [Accessed 15 August 2024].
- [156] Rather ZH, Pal BC. Online Estimation of Disturbance Size and Frequency Nadir Prediction in Renewable Energy Integrated Power Systems. IEEE Trans Power Syst 2023.
- [157] Zografos D, Rabuzin T, Ghandhari M, Eriksson R. Prediction of frequency nadir by employing a neural network approach. In: 2018 IEEE PES Innovative Smart Grid Technologies Conference Europe (ISGT-Europe): IEEE; 2018. p. 1–6.
- [158] Liu X, Xie J, Fang X, Yuan H, Wang B, Wu H, et al. A Comparison of Machine Learning Methods for Frequency Nadir Estimation in Power Systems. In: 2022 IEEE Kansas Power and Energy Conference (KPEC): IEEE; 2022. p. 1–5.
- [159] Panwar A, Rather ZH, Doolla S, Liebman A, Dargaville R. A Machine Learning Based Approach for Frequency Response Prediction in Low Inertia Power System. 2022 IEEE PES Innovative Smart Grid Technologies-Asia (ISGT. Asia): IEEE; 2022. p. 175–9.

- [160] de Sousa RP, Moreira C, Carvalho L, Matos M. In: Including Dynamic Security Constraints in Isolated Power Systems Unit Commitment/Economic Dispatch: a Machine Learning-based Approach. Belgrade PowerTech: IEEE; 2023. p. 1–6.
- [161] Sang L, Xu Y, Yi Z, Yang L, Long H, Sun H. Conservative Sparse Neural Network Embedded Frequency-Constrained Unit Commitment With Distributed Energy Resources. IEEE Trans Sustainable Energy 2023.
- [162] Zeng C, Qiu W, Wang W, Sun K, Chen C, Sundaresh L, et al. Disturbance Magnitude Estimation: MLP-based Fusion Approach for Bulk Power Systems. In: 2022 IEEE/IAS 58th Industrial and Commercial Power Systems Technical Conference (I&CPS): IEEE; 2022. p. 1–6.
- [163] Chiu H-W, Chang-Chien L-R. A supervised learning scheme for evaluating frequency nadir and fast response reserve in ancillary service market. IEEE Access 2021;9:100934–43.
- [164] Yuan H, Tan J, Zhang Y, Murthy S, You S, Li H, et al. Machine learning-based PV reserve determination strategy for frequency control on the WECC system. In: 2020 IEEE Power & Energy Society Innovative Smart Grid Technologies Conference (ISGT): IEEE; 2020. p. 1–5.
- [165] Oladeji I, Zamora R, Lie TT. An online security prediction and control framework for modern power grids. Energies 2021;14:6639.
- [166] Zhao Y, You S, Mandich M, Zhu L, Zhang C, Li H, et al. Deep learning-based adaptive remedial action scheme with security margin for renewable-dominated power grids. Energies 2021;14:6563.
- [167] Li G, Lou W, Ding L, Jin Z, Bao W, Terzija V. Equivalent Virtual Inertia Estimation of Wind Turbines Based on GS-LSTM. In: 2023 International Conference on Power System Technology (PowerCon): IEEE; 2023. p. 1–5.
- [168] Chen Y, Sun M, Chu Z, Camal S, Kariniotakis G, Teng F. Vulnerability and impact of machine learning-based inertia forecasting under cost-oriented data integrity attack. IEEE Trans Smart Grid 2022;14:2275–87.
- [169] Mihalič F, Truntič M, Hren A. Hardware-in-the-loop simulations: A historical overview of engineering challenges. Electronics 2022;11:2462.
- [170] Kumar Krishna A, Jaramillo-Cajica I, Auer S, Schiffer J. A power-hardware-in-theloop testbed for intelligent operation and control of low-inertia power systems. 2022.
- [171] Panda RK, Mohapatra A, Srivastava SC. Online estimation of system inertia in a power network utilizing synchrophasor measurements. IEEE Trans Power Syst 2019;35:3122–32.
- [172] Guo L, Ye J, Yang B. Cyberattack detection for electric vehicles using physicsguided machine learning. IEEE Trans Transp Electrif 2020;7:2010–22.
- [173] Opal RT. Autonomous vehicle, https://blob.opal-rt.com/medias/Mkt_0027406. pdf; 2023 [Accessed 15 August 2024].
- [174] Opal RT. Ampex Intelligent Systems Cybersecurity with Machine Learning and Modeling & Simulation, https://blob.opal-rt.com/medias/L00161_0934.pdf; 2022 [Accessed 15 August 2024].
- [175] Lin F-J, Chang C-F, Huang Y-C, Su T-M. A Deep Reinforcement Learning Method for Economic Power Dispatch of Microgrid in OPAL-RT Environment. Technologies 2023;11:96.
- [176] Roy B, Adhikari S, Datta S, Lalngaihawma S, Samanta S. Real-Time Simulation of a Wind-Solar-Battery Based Microgrid System. J Electr Eng Technol 2023;18: 2087–102.
- [177] Dantas DT, Li H, Charton T, Chen L, Zhang R. Machine learning based anomalybased intrusion detection system in a full digital substation. 2020.
- [178] Dharmapala KD, Rajapakse A, Narendra K, Zhang Y. Machine learning based realtime monitoring of long-term voltage stability using voltage stability indices. IEEE Access 2020;8:222544–55.
- [179] National instrument. Deploying deep learning models to NI hardware, https:// www.ni.com/en/shop/data-acquisition-and-control/add-ons-for-dataacquisition-and-control/what-is-vision-development-module/deploying-deeplearning-models-to-ni-hardware.html/; 2023 [Accessed 15 August 2024].
- [180] McClain C, Keller D. Applying Machine Learning Techniques in National Instruments' LabView to Identify NMR Signals. APS Division of Nuclear Physics. Meeting Abstracts 2020:p. FH. 003.
- [181] Speedgoat. Image processing, sensor fusion, deep learning, AI, and high-speed controls with Speedgoat low-latency vision I/O modules with vision protocol support, https://www.speedgoat.com/products-services/i-o-connectivity/ vision/; 2023 [Accessed 15 August 2024].
- [182] Kheshti M, Ding L, Bao W, Yin M, Wu Q, Terzija V. Toward intelligent inertial frequency participation of wind farms for the grid frequency control. IEEE Trans Ind Inf 2019;16:6772–86.
- [183] Veerakumar N, Cremer JL, Popov M. Dynamic Incremental Learning for real-time disturbance event classification. Int J Electr Power Energy Syst 2023;148:108988.